HEALTH TECHNOLOGY ASSESSMENT

VOLUME 23 ISSUE 28 JUNE 2019 ISSN 1366-5278

HPV testing compared with routine cytology in cervical screening: long-term follow-up of ARTISTIC RCT

Clare Gilham, Alexandra Sargent, Henry C Kitchener and Julian Peto



HPV testing compared with routine cytology in cervical screening: long-term follow-up of ARTISTIC RCT

Clare Gilham,¹ Alexandra Sargent,² Henry C Kitchener³ and Julian Peto¹*

¹Non-Communicable Disease Epidemiology Unit, London School of Hygiene & Tropical Medicine, London, UK

²Department of Virology, Central Manchester University Hospitals NHS Foundation Trust, Manchester, UK

³School of Cancer and Sciences, University of Manchester, St Mary's Hospital, Manchester, UK

*Corresponding author

Declared competing interests of authors: Henry Kitchener reports grants from the University of Manchester during the conduct of the study. He is chairperson of the Advisory Committee for Cervical Screening (Public Health England) as well as of the National HPV Pilot Steering Group. Any views expressed in this report are those of the authors and not of Public Health England.

Published June 2019 DOI: 10.3310/hta23280

This report should be referenced as follows:

Gilham C, Sargent A, Kitchener HC, Peto J. HPV testing compared with routine cytology in cervical screening: long-term follow-up of ARTISTIC RCT. *Health Technol Assess* 2019;**23**(28).

Health Technology Assessment is indexed and abstracted in Index Medicus/MEDLINE, Excerpta Medica/EMBASE, Science Citation Index Expanded (SciSearch®) and Current Contents®/ Clinical Medicine.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.513

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hta

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 98/04/501. The contractual start date was in September 2013. The draft report began editorial review in April 2017 and was accepted for publication in November 2017. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Honorary Professor, University of Manchester, and Senior Clinical Researcher and Associate Professor, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk

Abstract

HPV testing compared with routine cytology in cervical screening: long-term follow-up of ARTISTIC RCT

Clare Gilham,¹ Alexandra Sargent,² Henry C Kitchener³ and Julian Peto^{1*}

¹Non-Communicable Disease Epidemiology Unit, London School of Hygiene & Tropical Medicine, London, UK

²Department of Virology, Central Manchester University Hospitals NHS Foundation Trust, Manchester, UK

³School of Cancer and Sciences, University of Manchester, St Mary's Hospital, Manchester, UK

*Corresponding author julian.peto@lshtm.ac.uk

Background: The National Screening Committee (NSC) based its recommendation that human papillomavirus (HPV) testing should replace cytology in primary cervical screening largely on the 2009 follow-up results of the ARTISTIC trial (A Randomised Trial In Screening To Improve Cytology). The NSC must now decide on screening intervals and triage policy. Options include extending the screening interval up to 10 years for human papillomavirus-negative (HPV–) women, delaying recall for human papillomavirus-positive (HPV+) women with normal cytology (as their infections are usually transient), and basing triage on full HPV typing.

Methods: In ARTISTIC, 24,510 women were recruited who were attending routine cervical cytology in Greater Manchester in 2001–3. The women were randomly allocated between revealing and concealing their HPV test results and were recalled every 3 years. After 2009, the women returned to routine cytological screening with recall every 3 years for those aged < 50 years, and every 5 years for those aged 50–64 years. We have followed the cohort to 2015 through national cancer registration for CIN3 (cervical intraepithelial neoplasia grade 3) and cancer, and through linkage to the cervical screening call–recall system to obtain lifetime cytology records.

Results: The analysis comprised 24,496 women at round 1 and 13,591 women at round 2 (which was 30–48 months later). Follow-up via local histology laboratories and national cancer registration identified 505 cases of cervical intraepithelial neoplasia grade 3 or cervical cancer (CIN3+) (including 22 invasive cervical cancers). The cumulative CIN3+ risk 10 years after a negative HPV test [0.31%, 95% confidence interval (CI) 0.18% to 0.49%, in the revealed arm] was similar to that 3 years after negative cytology (0.30%, 95% CI 0.23% to 0.41%, in the concealed arm) and fell sharply with age, from 1.1% (95% CI 0.7% to 1.8%) in those women aged < 25 years to 0.08% (95% CI 0.03% to 0.20%) in those women aged > 50 years. The 10-year cumulative CIN3+ risk following a new HPV infection at round 2 was 3.4% (95% CI 2.1% to 5.4%). The highest risks were associated with type-specific persistent infections that, overall, resulted in a 10-year cumulative CIN3+ risk of 20.4% (95% CI 15.6% to 26.4%).

Conclusions: We found a similar level of protection 10 years after a negative HPV test and 3 years after negative cytology. These data support a considerably longer screening interval after a negative HPV test than after a negative cytology test. About three-quarters of women with HPV infection and normal cytology clear their infections within about 3 years. Their risk of CIN3+ within this time frame is low (1.5%), suggesting that the current policy of annual repeat testing and referral after 2 years may be unnecessarily cautious. Approximately 40% of women who remained HPV+ had cleared their initial infection and acquired

[©] Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: INIRH Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

a new HPV type. The cumulative CIN3+ risks in women with type-specific persistent infections are about six times higher than in women with new infections. Triage strategies based on HPV persistence would, therefore, reduce unnecessary referral of women with new (and largely transient) infections. HPV assays that identify HPV types 31, 33, 45, 52 and 58 in addition to 16 and 18 could be useful in triage as well as in primary HPV testing. Similar results in recent routine HPV screening suggest that our results are generalisable despite changes in cytology and HPV assay methods. We are continuing to follow the ARTISTIC cohort into the new era of primary HPV screening. Future work will focus on the implications of more sensitive HPV testing for primary HPV screening policy and triage of HPV-positive women. Our results suggest that a more sensitive test is needed to detect occult CIN3 at high risk of progression to cancer, but this would substantially increase the overall HPV detection rate. Tests such as DNA (deoxyribonucleic acid) methylation for distinguishing HPV infection from neoplasia will be evaluated on stored samples and on further samples now being collected from women in the cohort who are still being screened.

Funding: This project was funded by the National Institute for Health Research (NIHR) Health Technology Assessment programme and will be published in full in *Health Technology Assessment*; Vol. 23, No. 28. See the NIHR Journals Library website for further project information.

Contents

List of tables	ix
List of figures	xi
List of abbreviations	xiii
Plain English summary	xv
Scientific summary	xvii
Chapter 1 Introduction	1
Chapter 2 Objectives	3
Chapter 3 Methods Data collection during the trial (2001–9) Further data collection (2010–16) NHS Central Register for cancer incidence and mortality National screening programme call–recall cytology and human papillomavirus records Testing stored samples for human papillomavirus deoxyribonucleic acid	5 5 5 5 6
Chapter 4 Definitions and statistical methods Definition of round 2 follow-up Human papillomavirus clearance Human papillomavirus recurrence Type-specific human papillomavirus persistence Human papillomavirus status Calculation of cumulative risk for cervical intraepithelial neoplasia or cervical cancer Kaplan–Meier analysis of interval-censored data	7 7 7 7 7 8 8
 Chapter 5 Results Section 1: description of data Eligible data Natural history of human papillomavirus infection Comparison with data from the UK primary human papillomavirus screening pilot Invasive cervical cancer Section 2: the long-term protection of a negative human papillomavirus test and, hence, the safe screening interval at different ages Section 3: the role of human papillomavirus typing Section 4: triage of human papillomavirus-positive women, particularly the interval to retesting for human papillomavirus-positive women Section 5: optimal ages at starting and stopping screening 	 9 9 10 11 13 15 18 21 23
Chapter 6 Discussion Risk-based screening and cancer prevention The long-term protection of a negative human papillomavirus test and, hence, the safe screening interval at different ages The role of human papillomavirus typing	31 31 32 33

Triage and referral policy in relation to primary screening interval	33
Triage and referral policy in relation to type-specific human papillomavirus persistence	33
Triage of human papillomavirus-positive women	34
Triage: women with normal cytology	34
Triage: women with borderline/mild cytology	34
Ages at starting and stopping screening	35
Stopping screening	35
Age at starting screening	35
Study limitations	35
The effect of cervical sampling and biopsy on human papillomavirus clearance	35
Variation in cytology practice	35
Hybrid capture 2 test sensitivity	36
Human papillomavirus-negative cancers	36
Further research	36
Conclusions	36
Acknowledgements	39
References	41

List of tables

TABLE 1 The CIN3 and invasive cancer cases	10
TABLE 2Age-specific HRHPV prevalence at round 1 (entry) and round 2(30–48 months later)	11
TABLE 3Women who were HRHPV+ at round 2 by age at round 2. Results areshown separately for those HRHPV- at entry and those HRHPV+ at entry	12
TABLE 4 Clearance of new and persistent HRHPV infections	13
TABLE 5 The HPV type-specific recurrence following a negative HC2 test of HPVinfections that were detected at entry	14
TABLE 6 Cytology triage of HPV+ women in the UK primary HPV screening pilotsites and ARTISTIC rounds 1 and 2 by age	14
TABLE 7 Invasive cervical cancers identified through national registration data in24,946 women by HPV and cytology status at entry (follow-up to 2015)	15
TABLE 8National registrations of cervical cancer and CIN3 in ARTISTIC in2001–15 and in England in 2014	15
TABLE 9 Cumulative CIN3+ risks following a single HC2– HPV test at round 1 $(n = 20,687)$ and at round 2 $(n = 12,399)$ by age at test	16
TABLE 10 Cumulative CIN3+ risks by combination of results at round 1 and atround 2 (30–48 months after round 1)	19
TABLE 11 Cumulative CIN3+ risks following a single HPV test at entry byHPV genotype	20
TABLE 12Cumulative CIN3+ risks following round 2 for women with andwithout type-specific persistence of HPV infection since round 1	22
TABLE 13Cumulative CIN3+ risks following a single HPV test at round 1 by HPVtype and cytology	24
TABLE 14 Cumulative CIN3+ risks following round 2 in women who were HC2+ with negative cytology at entry ($n = 1120$) by HPV status at round 1 and round 2	26
TABLE 15Cumulative CIN3+ risks following round 2 in women with negativecytology at round 2 by HPV status at round 1 and round 2	27
TABLE 16Cumulative CIN3+ risks following round 2 in women with borderlineor mild cytology at round 2 by HPV status at round 1 and round 2	28
TABLE 17 Cumulative CIN3+ risks following round 2 by age in women with type-specific HPV persistence	29

List of figures

FIGURE 1 The CONSORT (Consolidated Standards of Reporting Trials) flow of diagram showing women followed up for cancer registration and mortality through national registration, and for cytology through call–recall cervical screening records	9
FIGURE 2 New infections: prevalence of HRHPV at round 2 in the ARTISTIC, NTCC (Italy) and POBASCAM (the Netherlands) trials in women who were HRHPV– at entry	12
FIGURE 3 New infections: prevalence (%) at round 2 of each HRHPV type in the ARTISTIC, NTCC and POBASCAM trials in women who were HRHPV– at entry	13
FIGURE 4 Cumulative CIN3+ risks (%) following a negative HPV test or normal cytology at entry	18
FIGURE 5 Cumulative CIN3+ risks by HPV detection from entry (483 CIN3, 23 invasive cervical cancers in 24,500 women)	21
FIGURE 6 Cumulative CIN3+ risks following round 2 for persistent and newly acquired HPV infection (82 CIN3+ in 13,591 women)	23
FIGURE 7 Cumulative CIN3+ risks following round 2 by age in women with type-specific persistence of any HRHPV	30
FIGURE 8 Age-specific HRHPV prevalence at round 2 by age at round 2 overall and in women who were HRHPV– at round 1, and cumulative 10-year CIN3+ risk by age at round 1 in women who were HC2– at round 1	31
FIGURE 9 Age-standardised prevalence (%) of CIN2+ during ARTISTIC by time since previous normal cytology to entry, and by time since cytologically normal entry to next test	32

List of abbreviations

ARTISTIC	A Randomised Trial In Screening To	HPV	human papillomavirus
	Improve Cytology		human papillomavirus negative
CI	confidence interval	HPV+	human papillomavirus positive
CIN	cervical intraepithelial neoplasia	HRHPV	high-risk human papillomavirus
CIN2+	cervical intraepithelial neoplasia grade 2 or worse	LBC	liquid-based cytology
CIN3+	cervical intraepithelial neoplasia	NHSCR	NHS Central Register
	grade 3 or cervical cancer	NIHR	National Institute for Health
DNA	deoxyribonucleic acid		Research
HC2	hybrid capture 2	NSC	National Screening Committee
HC2-	hybrid capture 2 negative	PCR	polymerase chain reaction
HC2+	hybrid capture 2 positive	RLU/Co	relative light units/mean control
	7		

Plain English summary

uman papillomavirus (HPV) causes cervical cancer. The latest scientific evidence shows that screening for HPV is better than screening for abnormal cytology with a 'smear' test, so HPV testing is being rolled out nationally. The main disadvantage is that more women will test positive and be referred for further tests. Most infections are harmless and clear without treatment, and balance must be achieved so that women who progress to CIN3 (pre-cancer) are identified but that unnecessary referral and anxiety for women is minimised.

A Randomised Trial In Screening To Improve Cytology (ARTISTIC) recruited 24,510 women attending for cervical screening in Greater Manchester in 2001–3. Cervical samples taken at recruitment and again at screening 3 and 6 years later were tested for HPV. The women then returned to routine screening. We have followed them through national screening records and cancer registration until the end of 2015.

By comparing the HPV results taken at entry with those collected 3 years later, we can categorise HPV infections into new and persistent. We have found that the CIN3 risk in women with persistent infections is about six times higher than in women with new infections, which in turn is about 30 times higher than in women with no infection.

About three-quarters of women with HPV infection but no abnormal cells clear their infections within 3 years. Their risk of pre-cancer within 3 years is low (1.5%) and so intensive follow-up is unnecessary. Moreover, 40% of those who remain human papillomavirus positive (HPV+) have cleared their initial infection and acquired a new infection, meaning that they are also at much lower risk of disease than those with a persistent infection. The current practice in the national pilot study of annual repeat testing and referral of anyone who is still HPV+ after 2 years may, therefore, be too conservative.

We have also shown that the CIN3 risk after 10 years in women testing negative for HPV is similar to the risk after about 3 years in women testing negative for cytology. This means that screening intervals could be extended for women testing negative for HPV.

Scientific summary

Background

The National Screening Committee (NSC) based its recommendation that human papillomavirus (HPV) testing should replace cytology in primary cervical screening largely on the 2009 follow-up results of A Randomised Trial In Screening To Improve Cytology (ARTISTIC) [URL: https://legacyscreening.phe.org.uk/ cervicalcancer (accessed 19 April 2018)]. The NSC must now decide on screening intervals in time for national roll-out of primary HPV screening, currently scheduled for December 2019. Options include extending the screening interval for up to 10 years for human papillomavirus-negative (HPV–) women and delaying recall for human papillomavirus-positive (HPV+) women by up to 3 years if their cytology is normal, and perhaps by only 1 year if their cytology is borderline or mild. HPV infections are usually transient and a substantial reduction in triage costs and procedures could be achieved if a longer delay in patient recall was shown to be safe.

Methods

In ARTISTIC, 24,510 women attending for routine cervical cytology in Greater Manchester were recruited between 2001 and 2003. Cytology was conducted as part of the national screening programme using liquid-based cytology (LBC) technology and women with abnormal cytology were managed under national guidelines, irrespective of HPV results. Women were recalled twice, 3 and 6 years after entry. LBC samples at entry and at rounds 2 and 3 were tested for HPV and the residual material stored. Women were randomly allocated to reveal or conceal their HPV test results, and on the revealed arm women with normal cytology who were HPV positive were recalled after 1 year for a repeat HPV test. Histology results were obtained from local laboratories. After 2009, follow-up and sample collection ended and the women returned to routine cytological screening with recall every 3 years for those aged < 50 years and every 5 years for those aged 50–64 years. We have followed the trial cohort through national cancer registration for cervical intraepithelial neoplasia grade 3 (CIN3) and cancer, and through linkage to the cervical screening call–recall system for lifetime cytology records. Cumulative cervical intraepithelial neoplasia grade 3 or cervical cancer (CIN3+) risks were calculated comparing women according to HPV and cytology status at baseline. Additional analyses began at round 2, when persistent and newly acquired infections could be distinguished and the high prevalence of accumulated CIN3 missed by earlier screening had been eliminated at entry.

Results

The analysis comprised 24,496 women at round 1 and 13,591 women at round 2 (defined as the first test 30–48 months later). Follow-up was via local histology laboratories until 2009 when the trial ended and then via cancer registration until April 2015. This identified 505 cases of CIN3+ (including 22 invasive cancers). Similar cumulative CIN3+ risks were seen 10 years after a negative hybrid capture 2 (HC2) test at entry [0.31%, 95% confidence interval (CI) 0.18% to 0.49%, in the revealed arm] and 3 years after a negative cytology test at entry (0.30%, 95% CI 0.23% to 0.41%, in the concealed arm). The 10-year cumulative CIN3+ risk in women who were hybrid capture 2 negative (HC2–) at entry was highest in women aged 20–24 years (1.10%, 95% CI 0.69% to 1.77%) and significantly higher (p < 0.001) in women aged 25–39 years (0.40%, 95% CI 0.28% to 0.56%) than in those aged > 40 years (0.11%, 95% CI 0.06% to 0.20%).

The availability of partial or full HPV genotyping assays allows risk-based stratification in an organised screening programme. Four out of the six sites in the UK pilot study are utilising partial typing HPV tests [Roche COBAS (Roche Molecular Diagnostics, Pleasanton, CA, USA) and Abbott Realtime Assays (Abbott Molecular,

[©] Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIRR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Maidenhead, UK)] that identify HPV 16 and HPV 18 infections. We found a much higher cumulative CIN3+ risk among women with HPV 16 infection than among women with any other genotype. Although HPV 16 constituted 22% of all hybrid capture 2-positive (HC2+) infections, 57% of CIN3+ occurred among this group of women, giving a 10-year cumulative CIN3+ risk of 29.8% (95% CI 26.8% to 33.0%). HPV 18 constituted 9% of all HC2+ infections (including 2% who also had HPV 16), with a 10-year cumulative CIN3+ risk similar to the group of high-risk HPV types comprising 31, 33, 45, 52 and 58.

The 10-year cumulative CIN3+ risk following a new high-risk HPV (HRHPV) infection at round 2 was low (3.4%, 95% CI 2.1% to 5.4%). HPV 16 again showed the highest 10-year risk following a new infection (7.3%, 95% CI 3.7% to 14.1%), suggesting that women with HPV 16 might be referred immediately. Much higher risks were associated with any type-specific persistence (the same HPV type as in round 1), which overall conferred a 10-year cumulative risk of 20.4% (95% CI 15.6% to 26.4%). The cumulative CIN3+ risk following type-specific persisting infection was similar regardless of age before the age of 40 years (23.8%, 95% CI 18.2% to 30.9%), but significantly lower (p = 0.02) in women aged \geq 40 years (6.8%, 95% CI 2.3% to 19.7%). Persistent HPV 16 infections account for 44% of type-specific infections in those aged < 30 years and only 11% among women aged \geq 40 years, but stratification by type does not entirely account for the reduction in CIN3+ risk in women aged \geq 40 years. Of the 331 women with double positive HRHPV tests, 115 (35%) were positive with a new HPV type at the second test and 216 (65%) were type-specific persistent. The proportion with new infections was highest in younger women (43% in women in their 20s) and decreased to 23% of double positive infections in women aged > 50 years. Entry samples for 17 out of the 23 cervical cancers diagnosed so far (including one diagnosed after the follow-up date) were HC2+. Five of the 6 HC2– entry samples were found to be HPV+ on retesting by PCR.

Conclusions

The CIN3+ risk 10 years after a negative HPV test is similar to that at 3 years after a negative cytology test. The risk at each age is approximately proportional to the incidence of new HPV infection in the population, which falls sharply with age and is very low in those aged \geq 40 years. These data support a longer screening interval after a negative HPV test than after a negative cytology test. This could be at least 5 years, and might be extended to 10 years for women aged \geq 50 years, or perhaps aged \geq 40 years. About three-quarters of women with HPV infection and normal cytology clear their infections within 3 years. Their risk of CIN3+ within this time is low (1.5%), suggesting that the protocol in the national pilot of annual repeat testing [Public Health England. HPV Primary Screening Pilot Protocol Algorithm. Version 3.0. Public Health England; 2016. URL: https://assets.publishing.service.gov.uk/government/uploads/system/ uploads/attachment data/file/529496/HPVPSFlowchart-Version3 Jan16.CURRENTppt.pdf (accessed 19 April 2018)] and referral after 2 years may be too conservative. Approximately 40% of women who remained high-risk HPV+ at round 2 had cleared their initial infection and acquired a different HPV type. They had less than 20% of the CIN3+ risk of those with type-specific persistence. Women with HPV 16 or HPV 18 and normal cytology are being referred to immediate colposcopy in some centres in the national pilot. Strategies based on full or partial HPV typing could be considered in triage as well as in primary HPV screening. Future work will focus on the implications of more sensitive HPV testing for primary HPV screening policy and triage of HPV-positive women. Our results suggest that a more sensitive test is needed to detect occult CIN3 at high risk of progression to cancer, but this would substantially increase the overall HPV detection rate. Tests such as DNA methylation for distinguishing HPV infection from neoplasia will be evaluated on stored samples and on further samples now being collected from women in the cohort who are still being screened.

Funding

Funding for this study was provided by the Health Technology Assessment programme of the National Institute for Health Research.

Chapter 1 Introduction

he NHS Cervical Screening Programme is already one of the most successful in the world in preventing cervical cancer,^{1,2} but unnecessary referral and treatment should be minimised to reduce NHS costs, inconvenience to women and unnecessary treatment for low-grade cervical dysplasia, which can compromise later birth outcomes. Human papillomavirus (HPV) testing is already used in the UK to triage borderline and mild cervical cytology and as a test of cure following excision of cervical intraepithelial neoplasia (CIN). National roll-out of primary HPV testing, scheduled to be complete by December 2019, will further increase sensitivity; however, referrals and, hence, costs will also increase without longer screening intervals and more efficient triage methods for human papillomavirus-positive (HPV+) women. Primary HPV testing with cytology triage of HPV+ women is being piloted at several sites in England following publication of ARTISTIC trial (A Randomised Trial In Screening To Improve Cytology) results over three rounds of HPV screening³⁻⁵ and pooled data from ARTISTIC and other randomised trials showing a reduction in long-term cervical cancer risk.⁶ The aim of primary HPV screening is to reduce cervical cancer risk by identifying women with high-grade CIN early enough to prevent progression to invasive cancer. The introduction of primary HPV testing in the NHS Cervical Screening Programme will require practical decisions on screening intervals for different age ranges as well as the triage protocol for HPV+ women. The results up to 2009 from the initial screening rounds in ARTISTIC contributed to the decision to pilot HPV primary screening,⁷ and this extension to 2015 gave a 10-year follow-up from entry, providing further evidence to optimise screening protocols.

The CIN grade 3 or cervical cancer (CIN3+) rate in the second round of ARTISTIC was less than half the rate at entry, as the majority of cases diagnosed at entry reflected a long-term accumulation missed by previous screening.⁵ Virtually all prevalent cases will be detected when HPV screening is first introduced. Therefore, follow-up from round 1 represents what will be seen when primary HPV screening is introduced, while follow-up from round 2 represents what will be seen at the second and subsequent rounds of HPV screening. For the purpose of modelling an established HPV screening programme, an important group are the 85% of women who were human papillomavirus negative (HPV–) at entry (round 1), as long follow-up of those with new infections detected at round 2 represents the time course of HPV persistence and CIN3+ risk following first infection. Extending follow-up to 2015 has increased the median follow-up beyond round 2 in ARTISTIC from 4 to 10 years.

Follow-up of the cohort to 2009 showed that the subsequent CIN3+ rate is substantially lower following a single negative HPV test than for negative cytology,⁴ and this was influential in the National Screening Committee's (NSC's) recommendation that HPV testing should replace cytology as the primary screening test. This further follow-up beginning at round 2 will show whether or not the screening interval can safely be increased from the current routine (3 years for those aged < 50 years and 5 years for those aged \geq 50 years) to up to 10 years for HPV– women, irrespective of previous test results.

Chapter 2 Objectives

The objective of this extended follow-up of the ARTISTIC cohort was to obtain the data required to evaluate the long-term benefits of alternative screening strategies using primary HPV testing, including the following:

- 1. The long-term cumulative CIN3+ risk following a negative HPV test, and, hence, the safe screening interval at different ages.
- The effect of HPV type on CIN3+ risk following a positive HPV test, and, hence, the role of HPV genotyping in routine screening.
- 3. The difference in CIN3+ risks following round 2 between women with type-specific HPV persistence since round 1 and those with a newly acquired HPV type. This is relevant to the triage protocol for HPV+ women: the interval to retesting, and either referral for colposcopic examination or further testing in women who remain HPV+ based on HPV typing and cytology.
- 4. The effects on CIN3+ risk of different ages at starting and stopping screening. The high prevalence of CIN2+ at age 20–24 years (6.3%) in round 1 of ARTISTIC⁴ showed that many cases detected when women are first screened at age 25 years developed more than 5 years earlier. The long-term CIN3 risk in HPV– women who are aged ≥ 50 years is very low, but if they are never screened again their lifetime then cancer risk may not be negligible.

Chapter 3 Methods

Data collection during the trial (2001–9)

ARTISTIC compared cytology with and without HPV testing among 24,510 women attending routine cervical screening in 2001–3. Over 60,000 HPV tests [hybrid capture 2 (HC2) with full HPV typing of those testing positive] were performed on routinely collected liquid-based cytology (LBC) cervical samples until September 2009. Methods are described elsewhere^{3,5} reporting that women were randomly allocated in a ratio of 1 : 3 between having their HPV results concealed (concealed arm, n = 6124) or revealed (revealed arm, n = 18,386). All women with abnormal cytology were recalled for retesting or referred to colposcopy according to national guidelines.³ After March 2008, the laboratory became part of the Sentinel Sites project in which low-grade cytological abnormalities were triaged using HPV. Women were referred with borderline or mild cytology only if they tested positive for HPV; otherwise, they were returned to routine recall. In addition, women in the revealed arm with normal cytology who were hybrid capture 2 positive (HC2+) were recalled for repeat HPV testing, and those whose HPV infection persisted were referred to colposcopy.

Liquid-based cytology was carried out using ThinPrep® (Hologic, Crawley, UK) and HPV testing using HC2 (Qiagen, Crawley, UK). A cut-off point of 1 relative light unit/mean control (RLU/Co) pg/µl was used to identify positive HPV samples, which were genotyped using at least one of three HPV typing assays. All HC2+ samples during rounds 1 and 2 were genotyped using the prototype Roche Line Blot Assay (Roche Molecular Diagnostics, Pleasanton, CA, USA).³ This was replaced by the commercially available Linear Array assay and the PapilloCheck® assay (Greiner Bio-one GmbH, Frickenhausen, Germany) during round 3. In addition, approximately two-thirds of archived HC2+ round 1 samples were also tested by the PapilloCheck assay and one-third of archived HC2+ round 2 samples were tested by the Roche Linear Array®. In all three rounds, any of the 13 high-risk human papillomavirus (HRHPV) types detected by HC2 (HPV types 16, 18, 31, 33, 35, 39, 45, 51, 52, 56, 58, 59, 68) that was detected by any of the assays was included in the analysis. All other HPV types were ignored in all analyses.

After the trial ended in 2009, all women returned to routine cytology every 3 years if they were aged < 50 years, and every 5 years if they were aged 50–64 years. We have published reports on baseline data,^{8,9} follow-up through two further screening rounds^{3–5,10} and various other HPV-related issues.^{6,11–13} These analyses were based on follow-up to 2009 through two cytology laboratories in the Manchester area. Any women moving out of the area were lost, and only 60% of the ARTISTIC cohort had a cytology record between 2006 and 2009.

Further data collection (2010–16)

NHS Central Register for cancer incidence and mortality

The cohort was flagged through NHS Central Register (NHSCR), giving notification of deaths and cancer diagnoses, including carcinoma in situ of the cervix (CIN3), until December 2015.

National screening programme call–recall cytology and human papillomavirus records

The cohort was linked to the NHS Cervical Screening Programme call–recall database to obtain lifetime cervical screening records for the entire ARTISTIC cohort. This included any recorded HPV results taken after triaging of borderline or mild dyskaryosis cytology or as part of primary HPV screening (in pilot areas). The reasons for ceasing screening, including hysterectomy, were also recorded.

[©] Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: INIRH Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Testing stored samples for human papillomavirus deoxyribonucleic acid

All samples taken during the trial have been tested using HC2 with HPV typing for HC2+ samples. Women who consented to their HPV samples being retained for future research are now stored at Professor Lorincz's laboratory at QMUL. Sensitive polymerase chain reaction (PCR)-based assays [Roche Line Blot assay and PapType[®] assay (Genera Biosystems, Scoresby, VA, Australia)] were carried out on hybrid capture 2-negative (HC2–) samples from women who have developed invasive cervical cancer since 2009.

Chapter 4 Definitions and statistical methods

Definition of round 2 follow-up

In previous reports, round 2 was defined as the first adequate cytology between 26 and 54 months after entry, which included 10% with no adequate HPV test on their round 2 sample. In this report, round 2 is defined as the first adequate HPV test between 30 and 48 months after entry. Under this definition, 13,591 women had a round 2 test, on average, 3 years and 7 months after entry. In analyses beginning at round 2, 185 women with CIN3+ registration/histology before this date and six women whose flagging data indicated that they had cancelled from the NHSCR before this date were excluded.

Human papillomavirus clearance

To estimate and compare clearance rates of recently acquired and long-standing HPV infections (i.e. those already present at entry), clearance from the second HPV test in ARTISTIC (regardless of time since entry) was assessed. Women with any histology, regardless of result, between the second and third HPV tests were excluded in case the action of taking a biopsy affected the clearance rate. In addition, women with CIN3+ histology before the second HPV test were excluded. A total of 1026 women with at least three HPV tests whose second test was positive for HRHPV were included in the analysis. Each infection with a HPV type was considered independently, so women with more than one HPV type were counted more than once in the analysis. Clearance of each infection was calculated by time between the second and third HPV test and stratified by whether or not the same HPV type was present at entry (i.e. whether the infection developed since entry or was prevalent at entry).

Human papillomavirus recurrence

The recurrence of type-specific HPV infections was counted in women who had HRHPV infections at round 1 and subsequently tested HC2–. The time to 'clearance' (defined as testing HC2–) was stratified into 'short term', defined as clearance of < 24 months after round 1, and 'long term', defined as clearance 24–54 months after round 1. Recurrence in these two groups of women was defined as any subsequent HRHPV+ result with one or more genotype identified that was also present at entry. Under the protocol, genotyping was not carried out on HC2– samples, which implies that 'clearance' means that the HPV infection had become undetectable by HC2.

Type-specific human papillomavirus persistence

The HPV types found in round 2 (30–48 months after entry) were compared with those identified in round 1 so that round 2 infections could be classified as new or persistent. Women with a type-specific persistent infection with or without a new infection of a different genotype were classified as persistent.

Human papillomavirus status

In most other analyses, women were classified hierarchically into mutually exclusive groups: HPV 16 or HPV 18, any of HPV types 31, 33, 45, 52 or 58 (without 16 or 18), any other HRHPV, HC2+ with no HRHPV, or HC2–.

[©] Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: INIRH Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Calculation of cumulative risk for cervical intraepithelial neoplasia or cervical cancer

Cumulative CIN3+ risks were estimated by Kaplan–Meier methods. Subgroups were compared using the log-rank test. Cumulative risks were calculated from entry (round 1) and also from round 2 when incident and persistent infections could be distinguished. Women were censored at date of last cytology prior to hysterectomy according to call–recall data. All analyses were censored on 30 April 2015 to allow for late registration of CIN3 and cancer.

Kaplan–Meier analysis of interval-censored data

The aim of the cumulative risk analyses is to estimate the probability that CIN3+ would be detected by a test at a given time after entry to follow-up, beginning either at entry to the trial or at round 2. This raises several issues.

- 1. We ignore the possibility that CIN3 occasionally regresses, so cumulative risks may overestimate the prevalence that would be observed if the next test were 5 or 10 years later.
- 2. CIN3+ histology is backdated to the beginning of follow-up if the histology/registration occurred within a year of the beginning of follow-up. These cases are shown at time zero giving an initial step in the Kaplan–Meier curve. Some of these cases were not present at the time of the initial abnormal cytology or positive HPV test but developed during the period of repeat testing.
- 3. Later CIN3+ histology/registration dates are backdated to the first test in the preceding year, then further backdated to the mid-point of the interval between that test and the preceding test. For three cancers and a single CIN3 lesion with no recorded cytology within a year, the histology date is taken as the end of the interval. One further cancer was censored on ceasing screening and thus excluded.

This modified Kaplan–Meier treatment of interval-censored data, in which lesions are assigned to the mid-point of the screening interval where they became detectable, gives results similar to a parametric analysis (Professor Peter Sasieni, Queen Mary University of London, 2017, personal communication). The second-order error is that the interval mid-point is not the expected value of the time when the CIN3 would be detectable, which may be slightly earlier or later depending on the assumed model of HPV acquisition and CIN3 development.

Chapter 5 Results

Section 1: description of data

Eligible data

The cohort that participated in ARTISTIC consisted of 24,510 women aged between 20 and 64 years and provided an adequate cytology and HPV test at enrolment. Fourteen women did not have a valid NHS number and so 24,496 women were flagged with NHS Digital for cancer registration and death. These women were then linked to the National Screening Programme call–recall system, which yielded a final cohort of 23,888 after excluding three non-matches and 605 type-2 opt-outs (those opting out of their data being accessed for medical research) (*Figure 1*). The call–recall data were linked by NHS number to ensure correct linkage. At least one cytology date and result from the laboratory database matched the call–recall data in all but 53 women (99.8%). These 53 women were individually examined and most had only one or two tests identified through the laboratory. We assumed that these cytology dates had not been recorded correctly on the call–recall system. For those whose call–recall screening records matched the Manchester laboratory data during the trial (2001–9), it was not uncommon for some records to be inconsistent between the systems.

Follow-up via local histology laboratories to 2009 when the trial ended and via cancer registrations to April 2015 identified 509 cases of CIN3+, including 23 invasive cancers. Three cases of CIN3 were excluded owing to censoring at last cytology as no link was made to call–recall data. In addition, one invasive cancer was excluded as the woman's screening record indicated that she had ceased screening after reaching the age of 65 years. Therefore, 505 cases of CIN3+ were included in the analysis (*Table 1*).



FIGURE 1 The CONSORT (Consolidated Standards of Reporting Trials) flow of diagram showing women followed up for cancer registration and mortality through national registration, and for cytology through call–recall cervical screening records.

Histology only (2001–9), <i>n</i> (%)	Registration only (2001–15), n (%)	Both histology and registration, <i>n</i> (%)	Total number
99 (20.4)	94ª (19.3)	293 (60.3)	486
	11 ^b (47.8)	12 (52.2)	23
61 (21.2)	7 (2.4)	220 (76.4)	288
21 (22.8)	11 (12.0)	60 (65.2)	92
17 (23.6)	30 (41.7)	25 (34.7)	72
0 (0.0)	57 (100.0)	0 (0.0)	57
99 (19.4)	105 (20.6)	305 (59.9)	509
	99 (20.4) 61 (21.2) 21 (22.8) 17 (23.6) 0 (0.0)	(2001–9), n (%) (2001–15), n (%) 99 (20.4) 94 ^a (19.3) 11 ^b (47.8) 11 ^b (47.8) 61 (21.2) 7 (2.4) 21 (22.8) 11 (12.0) 17 (23.6) 30 (41.7) 0 (0.0) 57 (100.0)	(2001–9), n (%)(2001–15), n (%)registration, n (%)99 (20.4)94° (19.3)293 (60.3)11° (47.8)12 (52.2)61 (21.2)7 (2.4)220 (76.4)21 (22.8)11 (12.0)60 (65.2)17 (23.6)30 (41.7)25 (34.7)0 (0.0)57 (100.0)0 (0.0)

TABLE 1 The CIN3 and invasive cancer cases

a Three CIN3 were excluded owing to censoring at last cytology (2007, n = 1; 2010, n = 2).

b One invasive cancer excluded owing to censoring at ceasing screening at age 64 years (after 2010).

Among women who had not ceased screening due to age by 2016 (n = 17,602), 642 women aged 30–64 years were recorded as ceased on the call–recall system due to hysterectomy: 0.3% aged 30–39 years, 2.5% aged 40–49 years, 5.3% aged 50–59 years and 6.9% aged 60–64 years.

Natural history of human papillomavirus infection

Table 2 shows the prevalence of HPV infection at rounds 1 and 2 by age. HC2 was the primary assay used and only HC2+ samples were further genotyped. Overall, 4% of samples tested positive by HC2 but no HRHPV was detected. These have been assumed to be HRHPV– in the analysis shown in *Table 3*, which shows prevalence of HRHPV by age at round 2 among women who were HPV– at round 1, and prevalence rates of type-specific persistence and new infection among those who were HPV– at round 1. The prevalence of new infection falls steeply with age in those women who were HPV– at round 1, from 19.7% at 20–24 years to 1.0% at 55–64 years. In contrast, among women who were HPV– at round 1, the proportion with persistence at round 2 (17.9%) shows little or no age trend. Round 2 HPV prevalence in women who were HPV– at entry in ARTISTIC and in the POBASCAM (the Netherlands)¹⁴ and NTCC (Italy) trials¹⁴ are shown by age in *Figure 2* and by HPV type in *Figure 3*, in which ARTISTIC data are restricted to the age range of the NTCC trial. The screening interval was 3 years in ARTISTIC and NTCC and 5 years in the POBASCAM trial. *Figure 2* shows remarkably similar HRHPV rates in England, Italy and the Netherlands above the age of 33 years, but higher rates in ARTISTIC for younger women. *Figure 3* shows the data by HPV genotype.

The proportion of HPV infections clearing increased by time interval to testing (*Table 4*). Clearance of new HPV infections (i.e. infections that had developed in the preceding screening interval) was higher than persisting HPV infections (i.e. those that were present at the previous screening round). Clearance of HPV 16 infection was lower than other infections and approximately 40% of new HPV 16 infections were still persisting in women tested over 3 years.

Short-term recurrence was assessed among 405 women who were HRHPV+ at entry and who tested HC2– less than 2 years after entry (*Table 5*). Of these, 27 (6.7%) women had a recurrent infection of the same HPV type present at entry. Long-term recurrence was assessed among 415 women who were HRHPV+ at entry and tested HC2– at least 2 years later. Of these, only five (1.2%) women had a later recurrence or reinfection of the HPV type present at entry.

	Prevalence, <i>n</i>	Prevalence, <i>n</i> (%)						
Age (years)	HPV 16/18	HPV 31/33/45/52/58	Other HRHPV	No HRHPV detected (HC2+)	HC2-	Total HRHPV+, n (%)		
Round 1 (entry	()							
20–24	408 (15.7)	254 (9.8)	204 (7.9)	167 (6.4)	1560 (60.2)	2593 (33.4)		
25–29	264 (10.2)	200 (7.7)	145 (5.6)	106 (4.1)	1874 (72.4)	2589 (23.5)		
30–34	190 (5.2)	200 (5.4)	131 (3.6)	164 (4.5)	2998 (81.4)	3683 (14.2)		
35–39	103 (2.6)	104 (2.6)	120 (3.1)	150 (3.8)	3462 (87.9)	3939 (8.3)		
40–44	58 (1.7)	56 (1.7)	70 (2.1)	128 (3.8)	3069 (90.8)	3381 (5.4)		
45–49	26 (1.0)	42 (1.6)	43 (1.6)	110 (4.1)	2496 (91.9)	2717 (4.1)		
50–54	25 (1.1)	29 (1.2)	28 (1.2)	90 (3.8)	2210 (92.8)	2382 (3.4)		
55–59	19 (1.0)	10 (0.5)	22 (1.1)	67 (3.4)	1853 (94.0)	1971 (2.6)		
60–64	11 (0.9)	7 (0.6)	11 (0.9)	47 (3.8)	1165 (93.9)	1241 (2.3)		
Total	1104 (4.5)	902 (3.7)	774 (3.2)	1029 (4.2)	20,687 (84.5)	24,496 (11.4)		
Round 2 (30–4	18 months)							
20–24	36 (9.4)	37 (9.6)	22 (5.7)	29 (7.6)	260 (67.7)	384 (24.7)		
25–29	73 (8.1)	57 (6.3)	33 (3.7)	69 (7.6)	673 (74.4)	905 (18.0)		
30–34	66 (4.7)	45 (3.2)	30 (2.2)	65 (4.7)	1187 (85.2)	1393 (10.1)		
35–39	25 (1.3)	51 (2.5)	40 (2.0)	73 (3.6)	1819 (90.6)	2008 (5.8)		
40–44	14 (0.6)	39 (1.7)	19 (0.8)	72 (3.2)	2129 (93.7)	2273 (3.2)		
45–49	19 (1.0)	23 (1.2)	11 (0.6)	48 (2.5)	1846 (94.8)	1947 (2.7)		
50–54	8 (0.5)	14 (0.8)	13 (0.8)	48 (2.9)	1574 (95.0)	1657 (2.1)		
55–59	9 (0.6)	6 (0.4)	8 (0.5)	32 (2.1)	1464 (96.4)	1519 (1.5)		
60–64	6 (0.6)	5 (0.5)	7 (0.6)	29 (2.6)	1053 (95.7)	1100 (1.6)		
65–69	0 (0.0)	0 (0.0)	3 (0.7)	8 (2.0)	394 (97.3)	405 (0.7)		
Total	256 (1.9)	277 (2.0)	186 (1.4)	473 (3.5)	12,399 (91.2)	13,591 (5.3)		

TABLE 2 Age-specific HRHPV prevalence at round 1 (entry) and round 2 (30–48 months later)

Comparison with data from the UK primary human papillomavirus screening pilot

A UK pilot study is trialling primary HPV screening in six sites in England. Women who are HPV+ and have borderline or worse cytology are referred to colposcopy and those with negative cytology are recalled after 12 months for repeat HPV testing and cytology.¹⁵ An immediate effect of introducing primary HPV testing is diagnosis of prevalent CIN3+ which had been missed by previous cytology. This was seen in ARTISTIC,¹⁰ the NTCC trial¹⁶ and in a German study,¹⁷ and a much lower CIN3 rate than at entry will presumably be seen in later screening rounds in the UK pilot study.¹⁵ *Table 6* shows that the prevalence of moderate/ severe cytology among HPV+ women in ARTISTIC was similar at each age in round 1 of ARTISTIC and in the pilot study, but in round 2 it was about three times lower and referral rates were halved. This reflects prevalent disease that had been missed by previous cytology but was reliably detected by HPV testing. When primary HPV testing has been rolled out nationally it may be safe to introduce a longer recall interval for HPV+ women with borderline/mild cytology in round 2 whose previous HPV test was negative.

[©] Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

	Infection stat	us at entry					
	HRHPV-		HRHPV+				
Age (years) at round 2	Number of women	New HRHPV at round 2, <i>n</i> (%)	Number of women	New HRHPV at round 2, <i>n</i> (%)	Persisting HRHPV type at round 2, <i>n</i> (%)		
20–24	239	47 (19.7)	145	20 (13.8)	28 (19.3)		
25–29	648	77 (11.9)	257	37 (14.4)	49 (19.1)		
30–34	1119	60 (5.4)	274	21 (7.7)	60 (21.9)		
35–39	1814	68 (3.8)	194	16 (8.2)	32 (16.5)		
40–44	2138	48 (2.3)	135	7 (5.2)	17 (12.6)		
45–49	1869	37 (2.0)	78	8 (10.3)	8 (10.3)		
50–54	1602	24 (1.5)	55	0 (0.0)	11 (20.0)		
55–59	1482	14 (0.9)	37	3 (8.1)	6 (16.2)		
60–64	1079	12 (1.1)	21	3 (14.3)	3 (14.3)		
65–69	396	1 (0.3)	9	0 (0)	2 (22.2)		
All women	12,386	388 (3.1)	1205	115 (9.5)	216 (17.9)		

TABLE 3 Women who were HRHPV+ at round 2 by age at round 2. Results are shown separately for those HRHPV- at entry and those HRHPV+ at entry

Persistence indicates the same HRHPV type at both rounds.







FIGURE 3 New infections: prevalence (%) at round 2 of each HRHPV type in the ARTISTIC, NTCC and POBASCAM trials in women who were HRHPV– at entry.

TABLE 4 Clearance of new and persistent HRHPV infections

	Proportion of infections clearing by time to next HPV test, <i>n/N</i> (%)							
Infections	< 1 year	1–2 years	2–3 years	\geq 3 years				
New HPV 16 infections	20/51 (39.2)	18/26 (69.2)	7/13 (53.9)	7/12 (58.3)				
Other new infections	142/249 (57.0)	107/144 (74.3)	53/65 (81.5)	66/74 (89.2)				
All new infections ^a	162/300 (54.0)	125/170 (73.5)	60/78 (76.9)	73/86 (84.9)				
HPV 16: present at entry	20/94 (21.3)	12/22 (54.6)	9/21 (42.9)	7/12 (58.3)				
Other infections: present at entry	91/243 (37.5)	52/104 (50.0)	44/65 (67.7)	36/48 (75.0)				
All infections present at entry ^b	111/337 (32.9)	64/126 (50.8)	53/86 (61.6)	43/60 (71.7)				

a Next HPV test following first detection of 634 HRHPV types in 504 women who did not have that HRHPV type at entry.b Second HPV test for 609 HRHPV types detected at entry in 522 women.

Invasive cervical cancer

Table 7 shows the 23 invasive cancers identified within the cohort between 2001 and 2015 by HPV result and cytology at entry. The six HC2– samples were further tested using PCR-based assays and five were found to contain HRHPV deoxyribonucleic acid (DNA) (HPV 16 in four women, HPV 51 in one woman). Low levels of HPV 52 DNA were detected in a later sample (3 years after entry but 3 years before cancer registration) taken from the sixth woman. The 10 prevalent cancers diagnosed within 7 months of entry were all PCR+ with moderate or worse cytology, and nine were HC2+. Test positivity at entry for the remaining 13 women whose cancers were diagnosed between 4 and 13 years later was 15% (2/13)

	Months from HC2– test to next HPV test							
Recurrence	0		12	24	36	≥ 48		
Short-term recurrence (following a HC2- test < 24 months after entry)								
Number of women	26	81	122	118	29	29		
Cumulative number of women	26	107	229	347	376	405		
Cumulative recurrent infections (%)	2 (7.69)	7 (6.54)	14 (6.11)	21 (6.05)	22 (5.85)	27 (6.67)		
Long-term recurrence (following a HC2- t	est 24–54 mo	onths after en	itry)					
Number of women	11	54	81	77	175	17		
Cumulative number of women	11	65	146	223	398	415		
Cumulative recurrent infections (%)	0 (0.0)	0 (0.0)	1 (0.68)	1 (0.45)	5 (1.26)	5 (1.20)		

TABLE 5 The HPV type-specific recurrence following a negative HC2 test of HPV infections that were detected at entry

TABLE 6 Cytology triage of HPV+ women in the UK primary HPV screening pilot sites and ARTISTIC rounds 1 and 2 by age

		en, <i>n</i> (%)			
Age (years) at test	HPV+,ª <i>n</i> (%)	Negative	Borderline/mild	Moderate+	Referral rate, ^b %
Pilot sites					
25–29	8231 (28.3)	5108 (62.1)	2019 (24.5)	1104 (13.4)	11.0
30–39	6127 (13.6)	4170 (68.1)	1330 (21.7)	627 (10.2)	4.5
40–49	4024 (8.0)	2951 (73.3)	822 (20.4)	251 (6.2)	2.2
50–59	1937 (6.0)	1453 (75.0)	365 (18.8)	119 (6.1)	1.5
60–64	333 (4.7)	273 (82.0)	42 (12.6)	18 (5.4)	0.9
All women	20,652 (12.6)	13,955 (67.6)	4578 (22.2)	2119 (10.3)	4.2
Round 1 ARTISTIC					
25–29	716 (27.7)	374 (52.2)	245 (34.4)	96 (13.4)	13.2
30–39	1161 (15.2)	679 (58.5)	339 (29.1)	144 (12.4)	6.3
40–49	533 (8.7)	343 (64.4)	137 (25.7)	53 (9.9)	3.1
50–59	290 (6.7)	238 (82.1)	40 (13.8)	12 (4.1)	1.2
60–64	76 (6.1)	68 (89.5)	3 (3.9)	5 (6.6)	0.6
All women	2776 (12.7)	1702 (61.3)	764 (27.5)	310 (11.2)	4.9
Round 2 ARTISTIC					
25–29	230 (25.6)	160 (69.6)	63 (27.4)	7 (3.0)	7.8
30–39	393 (11.6)	267 (67.9)	107 (27.2)	19 (4.8)	3.7
40–49	239 (5.7)	193 (80.8)	40 (16.7)	6 (2.5)	1.1
50–59	137 (4.3)	113 (82.5)	23 (16.8)	1 (0.7)	0.8
60–64	46 (4.2)	43 (93.5)	3 (6.5)	0 (0.0)	0.3
All women	1045 (8.2)	776 (74.3)	236 (22.6)	33 (3.2)	2.1

a HC2 assay used in ARTISTIC and either Gen-Probe Aptima® (Gen-Probe Incorp., San Diego, CA, USA), Abbott Realtime® (Abbott Molecular, Maiden head, UK) or Roche COBAS® (Roche Molecular Diagnostics, Pleasanton, CA, USA) assays used in the pilot sites.

b Women who are HPV+ and have abnormal cytology are immediately referred to colposcopy according to the pilot sites protocol.

	Cytology		HC2+		HC2-			
Time from entry to cancer registration	Normal n (%)	Abnormal n (%)	PCR+ n (%)	PCR– n (%)	PCR+ n (%)	PCR– n (%)	Total	
0–7 months	0	10 (0.3)	9 ^a (0.3)	0	1 (0.6)	0	10	
8–47 months	0	0	0	0	0	0	0	
4–13 years	11 (0.05)	2 (0.06)	8 (0.3)	0	4 (2.5)	1 ^b (0.005)	13	
Number of women	21,369	3127	2780	1029	157°	20,530	24,496	

 TABLE 7 Invasive cervical cancers identified through national registration data in 24,946 women by HPV and cytology status at entry (follow-up to 2015)

a The entry sample was HC2+ with glandular neoplasia but insufficient for further testing in a woman whose cancer was diagnosed 4 months after entry. As the remaining 16 HC2+ samples were all PCR+ she is shown as PCR+.

b Low-level HPV52 DNA was detected in a later HC2– sample taken 3 years after entry and 3 years before cancer registration from the only HC2–PCR– woman.

c Estimated from PapilloCheck results at round 3 (0.76% HPV+ in 4473 HC2– samples).¹³ Entry samples for women who developed cancer were tested later with RLB, but PCR was not done on other HC2– entry samples.

for cytology, 62% (8/13) for HC2 and 92% (12/13) for PCR. *Table 8* shows that cancer is almost as common as CIN3 in those women aged \geq 55 years, indicating the need to focus on sensitivity for cancer as well as CIN3 detection in a woman's final (exit) screen. With up to 14 years of follow-up since trial entry in 2001–3, there have so far been no cervical cancer deaths in the cohort.

Section 2: the long-term protection of a negative human papillomavirus test and, hence, the safe screening interval at different ages

Table 9 shows cumulative CIN3+ risks in women testing HC2– at rounds 1 and 2 by age. The 10-year risks decrease from 1.10% [95% confidence interval (CI) 0.69% to 1.77%] in women aged 20 –25 years at entry to 0.08% (95% CI 0.03% to 0.20%) in women aged \geq 50 years at entry. The 10-year CIN3+ risk in all HPV– women aged 25–39 years was 0.40% (95% CI 0.28% to 0.56%) and significantly higher than the risk in women aged \geq 40 years at entry (0.11%, 95% CI 0.06% to 0.20%; *p* < 0.0001). Table 10 shows cumulative risks from round 2 in those HC2– at both time points compared with those women who had cleared an infection present at entry. Risks were based on very small numbers, but were approximately

	Age at registration (years)						
	20–29	30–39	40–49	50–59	60–69		
ARTISTIC							
Cervical cancer	0	7	10	4	2		
CIN3	158	163	55	10	2		
CIN3-to-cancer ratio	∞	23.3	5.5	2.5	1.0		
England							
Cervical cancer	483	643	521	330	230		
CIN3	14,458	7752	2692	848	206		
CIN3-to-cancer ratio	29.9	12.1	5.2	2.6	0.9		

TABLE 8 National registrations of cervical cancer and CIN3 in ARTISTIC in 2001–15 and in England in 2014

Age (years) at round 1	Number (%) of women at baseline	Number of CIN3+ women ^a	Risk from round 1						
			2.5-year		5-year		10-year		
			Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI	
20–24	1563	19	3	0.19 (0.06 to 0.59)	6	0.39 (0.17 to 0.86)	17	1.10 (0.69 to 1.77)	
25–29	1870	16	5	0.27 (0.11 to 0.64)	8	0.43 (0.22 to 0.86)	14	0.76 (0.45 to 1.28)	
30–39	6466	20	7	0.11 (0.05 to 0.23)	12	0.19 (0.11 to 0.33)	19	0.30 (0.19 to 0.46)	
40–49	5562	9	4	0.07 (0.03 to 0.19)	7	0.13 (0.06 to 0.26)	8	0.14 (0.07 to 0.29)	
50–64	5226	4	2	0.04 (0.01 to 0.15)	4	0.08 (0.03 to 0.20)	4	0.08 (0.03 to 0.20)	
All women (HC2–)	20,687 (84.5)	68	21	0.10 (0.07 to 0.16)	37	0.18 (0.13 to 0.25)	62	0.31 (0.24 to 0.39)	

TABLE 9 Cumulative CIN3+ risks following a single HC2– HPV test at round 1 (n = 20,687) and at round 2 (n = 12,399) by age at test

			Risk from round 1									
Age (years) at round 2	Number (%) of women at round 2	Number of CIN3+ women	2.5-year		5-year		10-year					
			Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI				
23–24	304	1	0		1	0.33 (0.05 to 2.35)	1	0.33 (0.05 to 2.35)				
25–29	701	0	0		0		0					
30–39	3131	9	4	0.13 (0.05 to 0.34)	5	0.16 (0.07 to 0.38)	9	0.29 (0.15 to 0.56)				
40–49	3949	2	1	0.03 (0.00 to 0.18)	1	0.03 (0.00 to 0.18)	2	0.05 (0.01 to 0.20)				
50–68	4314	1	1	0.02 (0.00 to 0.16)	1	0.02 (0.00 to 0.16)	1	0.02 (0.00 to 0.16)				
All women (HC2–)	12,399 (91.2)	13	6	0.05 (0.02 to 0.11)	8	0.07 (0.03 to 0.13)	13	0.11 (0.06 to 0.2)				
			Risk from round 1	Risk from round 1								
---	------	----------	------------------------------	---------------------	------------------------------	---------------------	------------------------------	--------------------	--	--	--	--
Number (% Age (years) at of women round 1 at baseline		Number	2.5-year		5-year		10-year					
		of CIN3+	Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI				
20–24	489	11	4	0.82 (0.31 to 2.17)	9	1.85 (0.97 to 3.53)	11	2.27 (1.26 to 4.06				
25–29	523	9	2	0.38 (0.10 to 1.52)	4	0.77 (0.29 to 2.03)	8	1.55 (0.78 to 3.08				
30–39	1642	13	6	0.37 (0.16 to 0.81)	8	0.49 (0.25 to 0.98)	11	0.68 (0.37 to 1.22				
40–49	1340	3	0		1	0.07 (0.01 to 0.53)	2	0.15 (0.04 to 0.60				
50–64	1341	2	1	0.07 (0.01 to 0.53)	2	0.15 (0.04 to 0.60)	2	0.15 (0.04 to 0.60				
All women (cytology negative)	5335	38	13	0.24 (0.14 to 0.42)	24	0.45 (0.30 to 0.67)	34	0.65 (0.46 to 0.90				

TABLE 9 Cumulative CIN3+ risks following a single HC2- HPV test at round 1 (n = 20,687) and at round 2 (n = 12,399) by age at test (continued)

a In 12 years of follow-up.

three times higher among those women who cleared an infection at entry. Presumably this is confounded with sexual activity and, hence, risk of acquiring a new HPV infection. *Figure 4* and the bottom panel of *Table 9* show a similar CIN3+ risk 10 years after a negative HPV test (0.31%, 95% CI 0.18% to 0.49% in the revealed arm) and 3 years after a negative cytology test (0.30%, 95% CI 0.23% to 0.41% in the concealed arm). These data support a much longer screening interval after a negative HPV test than after a negative cytology test.

Section 3: the role of human papillomavirus typing

The increasing availability of partial genotyping allows risk-based stratification in an organised screening programme. Four of the six sites are utilising partial typing HPV tests (Roche COBAS and Abbott Realtime assays) in the English pilot study, which both allow identification of HPV 16 and HPV 18 infections. *Table 11* and *Figure 5* show that the cumulative risk for women with HPV 16 is much higher than for any other genotype. Women with HPV 16 constituted 22% of all HC2+ infections, but 57% of CIN3+ occurred among this group of women, giving a 10-year cumulative CIN3+ risk of 29.8% (95% CI 26.8% to 33.0%). HPV 18 was a much less common type among ARTISTIC women, constituting 9% of all HC2+ infections (including 2% who also had HPV 16) and the 10-year cumulative risk did not differ from the group of types including 31, 33, 45, 52 and 58. It is clear that assays that could identify these five types in addition to HPV 16 and 18 would be beneficial for a risk-based referral strategy.

Table 12 and Figure 6 show the risks following two tests approximately 3 years apart. The 10-year cumulative CIN3+ risk following a new HPV infection is low at 3.4% (95% CI 2.1% to 5.4%). The 10-year CIN3 risk from round 2 was no higher in those women who were HPV+ with another type at round 1 (2.7%, 95% CI 0.9% to 8.0%) than in those women who were HPV- at round 1 (3.6%, 95% CI 2.2% to 6.0%).

The highest risks were associated with type-specific persistent infections that, overall, conferred a 10-year cumulative risk of 20.4% (95% CI 15.65% to 26.45%). HPV 16 again showed the highest risks and the 10-year risk following a new infection reached 7.3% (95% CI 3.7% to 14.1%), suggesting that any women with HPV 16 detected could be referred immediately. Of the 331 women with double-positive HRHPV tests



FIGURE 4 Cumulative CIN3+ risk (%) following a negative HPV test or normal cytology at entry.

	women at		Risk from round 2							
		Number	2.5-year		5-year		10-year			
HPV status at rounds 1 and 2		of CIN3+ women	Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI		
HC2 results (aged 20–64 yea	rs at entry)									
HC2 double positive	510	54	45	8.8 (6.7 to 11.6)	49	9.6 (7.4 to 12.5)	54	10.6 (8.2 to 13.6)		
HC2 negative positive	682	15	12	1.8 (1.0 to 3.1)	14	2.1 (1.2 to 3.5)	15	2.2 (1.3 to 3.6)		
HC2 positive negative	1249	3	1	0.08 (0.01 to 0.57)	2	0.16 (0.04 to 0.65)	3	0.25 (0.08 to 0.76)		
HC2 double negative	11,150	10	5	0.04 (0.02 to 0.11)	6	0.05 (0.02 to 0.112)	10	0.09 (0.05 to 0.17)		
Aged 25–39 years at entry										
HC2 positive negative	615	3	1	0.16 (0.02 to 1.15)	2	0.33 (0.08 to 1.30)	3	0.49 (0.16 to 1.52)		
HC2 double negative	4019	6	3	0.07 (0.02 to 0.23)	3	0.07 (0.02 to 0.23)	6	0.15 (0.07 to 0.33)		
Aged 40–64 years at entry										
HC2 positive negative	410	0	0		0		0			
HC2 double negative	6673	3	2	0.03 (0.01 to 0.12)	2	0.03 (0.01 to 0.12)	3	0.05 (0.02 to 0.15)		

TABLE 10 Cumulative CIN3+ risks by combination of results at round 1 and at round 2 (30-48 months after round 1)

Risk from round 1 10-year 2.5-year 5-year Number (%) Single HPV test of women at Number of CIN3+ Number of CIN3+ Number of CIN3+ of CIN3+ 95% CI 95% CI 95% CI HPV 16 831 (3.4) 249 219 26.4 (23.5 to 29.5) 238 28.7 (25.7 to 31.9) 247 29.8 (26.8 to 33.0) 13.2 (9.7 to 17.9) HPV 18 273 (1.1) 36 8.4 (5.7 to 12.4) 23 32 11.7 (8.4 to 16.2) 36 HPV 31/33/45/ 902 (3.7) 104 75 8.3 (6.7 to 10.3) 89 9.9 (8.1 to 12.0) 101 11.3 (9.4 to 13.5) 52/58 Other HRHPV 774 (3.2) 2.6 (1.7 to 4.0) 3.4 (2.3 to 4.9) 3.8 (2.6 to 5.4) 30 20 26 29 HC2+ non-high 1029 (4.2) 18 8 0.8 (0.4 to 1.6) 13 1.3 (0.7 to 2.2) 18 1.8 (1.1 to 2.8) risk^b 11.4 (10.4 to 12.4) All HC2+ 9.1 (8.2 to 10.0) 3809 (15.5) 437 345 398 10.5 (9.5 to 11.5) 431 21 HC2-20,687 (84.5) 68 0.10 (0.07 to 0.16) 37 0.18 (0.13 to 0.25) 62 0.31 (0.24 to 0.39) Total women 24,496 (100) 505 366 435 493

TABLE 11 Cumulative CIN3+ risks following a single HPV test at entry by HPV genotype

a In 12 years of follow-up.

b HC2+ but no HRHPV detected, including 27 women at entry with insufficient samples for typing (in whom two CIN3+ cases were diagnosed).



FIGURE 5 Cumulative CIN3+ risks by HPV detection from entry (483 CIN3, 23 invasive cervical cancers in 24,500 women).

(see *Table 3*), 115 (35%) were positive with a new HPV type at the second test and 216 (65%) was type-specific persistent. The proportion of new infections was highest in younger women (43% in women in their 20s) and decreased to just 23% of double-positive infections in women aged \geq 50 years.

Section 4: triage of human papillomavirus-positive women, particularly the interval to retesting for human papillomavirus-positive women

Table 13 shows cumulative CIN3+ risks by HPV type and cytology at entry. Referring moderate or worse cytology to immediate colposcopy is clearly a highly effective strategy for identifying those women at highest risk of disease (52% by 5 years). Among HC2+ women, the 10-year cumulative CIN3+ risks in those with negative and borderline/mild cytology were estimated to be 4.1% (95% CI 3.3% to 5.0%) and 10.3% (95% CI 8.7% to 12.2%), respectively. These risks vary by HPV genotype, and, hence, a partial genotyping assay could identify those at highest risk (e.g. HPV 16 or 18). Table 14 is restricted to the 1116 women who were HC2+ and cytology negative and had a follow-up sample taken 30–48 months later (round 2). The current UK protocol for HC2+ cytology-negative women is to repeat HPV and cytology testing at 12 and 24 months and refer all those still HPV+ at 24 months. Table 14 shows these women under a protocol of repeat testing after a 3-year recall interval. This excludes 10 CIN3+ that were diagnosed in cytology-negative HPV+ women in the revealed arm who were diagnosed before the round 2 test, but under a 3-year recall protocol would probably have been diagnosed at round 2. Analysis restricted to the concealed arm would avoid this bias. Only 277 out of the 1120 women in Table 14 were in the concealed arm and the results were consistent between the arms; therefore, they were pooled to maximise numbers. The majority of women cleared their infection by round 2, with 72% testing HC2– and a further 7% who remained HC2+ in whom no HRHPV was detected. A small proportion (1.3%) of women developed moderate or worse cytology by round 2, most (9 out of 15) of whom had CIN3+ immediately diagnosed. Out of 220 women who remained HPV+ with normal, borderline or mild cytology, 38% were infected with a new HPV type and 62% were type-specific persistent. Among the new infections, the 10-year cumulative CIN3+ risk remained low (3.7%, 95% CI 1.2% to 11.0%). The risks were slightly higher for those women with borderline or mild cytology, but there was insufficient power to detect a significant difference. Further genotyping of these samples to determine type-specific persistence would refer only those women at highest risk of developing CIN3+ to colposcopy.

[©] Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

			Risk from round 2							
	N		2.5-year		5-year		10-year			
HPV status at round 2	Number of women at round 2	n CIN3+	n CIN3+ (%)	95% CI	n CIN3+ (%)	95% CI	n CIN3+ (%)	95% CI		
Women with type-specific pers	istence									
Persistent HPV 16	66	22	20	30.3 (20.7 to 42.9)	21	31.8 (22.1 to 44.5)	22	33.4 (23.4 to 46.1		
Persistent HPV 18	23	3	3	13.0 (4.4 to 35.2)	3	13.0 (4.4 to 35.2)	3	13.0 (4.4 to 35.2)		
Persistent 31/33/45/52/58	83	15	12	14.5 (8.5 to 24.1)	13	15.7 (9.4 to 25.4)	15	18.1 (11.3 to 28.2		
Persistent other HRHPV	44	4	3	6.8 (2.3 to 19.7)	4	9.1 (3.5 to 22.4)	4	9.1 (3.5 to 22.4)		
Any HRHPV type specific persistence	216	44	38	17.6 (13.1 to 23.4)	41	19.0 (14.4 to 24.9)	44	20.4 (15.6 to 26.4		
Women without type-specific p	persistence									
New HPV 16	109	8	7	6.4 (3.1 to 13.0)	8	7.3 (3.7 to 14.1)	8	7.3 (3.7 to 14.1)		
New HPV 18	50	0	0	0.0	0	0.0	0	0.0		
New 31/33/45/52/58	193	8	7	3.6 (1.8 to 7.5)	7	3.6 (1.8 to 7.5)	8	4.2 (2.1 to 8.1)		
New other HRHPV	151	1	0	0.0	0	0.0	1	0.7 (0.1 to 4.8)		
Any new HRHPV	503ª	17	14	2.8 (1.7 to 4.7)	15	3.0 (1.8 to 4.9)	17	3.4 (2.1 to 5.4)		
HC2+ but no HRHPV	473	8	5	1.1 (0.4 to 2.5)	7	1.5 (0.7 to 3.1)	8	1.7 (0.9 to 3.4)		
HC2–	12,399	13	6	0.05 (0.02 to 0.11)	8	0.07 (0.03 to 0.13)	13	0.11 (0.06 to 0.19		
HC2 results ignoring genotypin	Ig									
HC2 double positive	510	54	45	8.8 (6.7 to 11.6)	49	9.6 (7.4 to 12.5)	54	10.6 (8.2 to 13.6)		
HC2 negative positive	682	15	12	1.8 (1.0 to 3.1)	14	2.1 (1.2 to 3.5)	15	2.2 (1.3 to 3.6)		
HC2 positive negative	1249	3	1	0.08 (0.01 to 0.57)	2	0.16 (0.04 to 0.65)	3	0.25 (0.08 to 0.76		
HC2 double negative	11,150	10	5	0.04 (0.02 to 0.11)	6	0.05 (0.02 to 0.112)	10	0.09 (0.05 to 0.17		
All women	13,591	82	63		71		82			

TABLE 12 Cumulative CIN3+ risks following round 2 for women with and without type-specific persistence of HPV infection since round 1



FIGURE 6 Cumulative CIN3+ risks following round 2 for persistent and newly acquired HPV infection (82 CIN3+ in 13,591 women).

Tables 15 and *16* show cumulative CIN3+ risks following new and persisting HPV infections at round 2 in women with negative and borderline/mild cytology, respectively. Similar patterns are seen, with the highest risks in women with persisting infections and in those with HPV 16 infections.

Section 5: optimal ages at starting and stopping screening

Table 17 and Figure 7 show that the cumulative CIN3+ risk 5 years after round 2 in women with type-specific persisting infection is similar regardless of age before the age of 40 years (23.8%, 95% CI 18.2% to 30.9%), but significantly lower in women aged \geq 40 years (6.8%, 95% CI 2.3% to 19.7%; p = 0.02). Persistent HPV 16 infections account for 44% of the type-specific infections in those women aged < 30 years, but only 11% among women aged \geq 40 years.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

		Number of CIN3+ women ^ª	Risk from round 1							
	Number (%)		2.5-year		5-year		10-year			
Negative cytology of	of women at baseline		Number of CIN3+ women (%)	95% Cl	Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI		
HPV 16	332 (1.6)	41	22	6.6 (4.4 to 9.9)	36	10.9 (8.0 to 14.7)	41	12.4 (9.3 to 16.5)		
HPV 18	146 (0.7)	10	1	0.7 (0.1 to 4.8)	6	4.1 (1.9 to 8.9)	10	6.9 (3.8 to 12.4)		
HPV 31/33/45/52/58	482 (2.3)	25	6	1.3 (0.6 to 2.8)	14	2.9 (1.7 to 4.9)	22	4.6 (3.1 to 7.0)		
Other HRHPV	479 (2.2)	9	3	0.6 (0.2 to 1.9)	7	1.5 (0.7 to 3.1)	8	1.7 (0.8 to 3.3)		
HC2+ non-high risk ^ь	784 (3.7)	9	1	0.1 (0.02 to 0.90)	5	0.6 (0.3 to 1.5)	9	1.2 (0.6 to 2.2)		
All HC2+	2223 (10.4)	94	33	1.5 (1.1 to 2.1)	68	3.1 (2.4 to 3.9)	90	4.1 (3.3 to 5.0)		
HC2–	19,146 (89.6)	53	11	0.06 (0.03 to 0.10)	25	0.13 (0.09 to 0.19)	48	0.26 (0.19 to 0.34)		
All women with	21,369 (100)	147	44		93		138			

TABLE 13 Cumulative CIN3+ risks following a single HPV test at round 1 by HPV type and cytology

negative cytology

		Number of CIN3+ women ^a	Risk from round 1							
Borderline/mild cytology at round 1	Number (%)		2.5-year		5-year		10-year			
	of women at baseline		Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI		
HPV 16	283 (10.6)	65	54	19.1 (15.0 to 24.0)	59	20.9 (16.6 to 26.1)	63	22.3 (17.9 to 27.6)		
HPV 18	93 (3.5)	10	7	7.5 (3.7 to 15.1)	10	10.8 (5.9 to 19.1)	10	10.8 (6.0 to 19.1)		
HPV 31/33/45/52/58	309 (11.6)	31	24	7.8 (5.3 to 11.4)	29	9.4 (6.6 to 13.2)	31	10.1 (7.2 to 14.0)		
Other HRHPV	259 (9.7)	10	7	2.7 (1.3 to 5.6)	9	3.5 (1.8 to 6.6)	10	3.9 (2.1 to 7.1)		
HC2+ non-high risk ^b	226 (8.5)	6	4	1.8 (0.7 to 4.7)	5	2.2 (0.1 to 5.2)	6	2.7 (1.2 to 5.8)		
All HC2+	1170 (43.9)	122	96	8.2 (6.8 to 9.9)	112	9.6 (8.0 to 11.4)	120	10.3 (8.7 to 12.2)		
HC2-	1495 (56.1)	10	6	0.4 (0.2 to 0.9)	7	0.5 (0.2 to 1.0)	9	0.6 (0.3 to 1.2)		
All women with borderline/mild cytology	2665 (100)	132	102		119		129			

NIHR Journals Library www.journalslibrary.nihr.ac.uk

HEALTH TECHNOLOGY ASSESSMENT 2019 VOL. 23 NO. 28

TABLE 13 Cumulative CIN3+ risks following a single HPV test at round 1 by HPV type and cytology (continued)

			Risk from round 1							
Moderate/severe cytology at round 1	Number (%)	Number	2.5-year		5-year		10-year			
	of women at baseline	of CIN3+ women ^a	Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI		
HPV 16	216 (46.8)	143	143	66.2 (59.9 to 72.4)	143	66.2 (59.9 to 72.4)	143	66.2 (59.9 to 72.4)		
HPV 18	34 (7.4)	16	15	44.1 (29.4 to 62.2)	16	47.1 (32.1 to 64.9)	16	47.1 (32.1 to 64.9)		
HPV 31/33/45/52/58	111 (24.0)	48	45	40.5 (32.1 to 58.3)	46	41.4 (32.9 to 51.2)	48	43.2 (34.6 to 53.0)		
Other HRHPV	36 (7.8)	11	10	27.8 (16.0 to 45.5)	10	27.8 (16.0 to 45.5)	11	30.8 (18.4 to 48.7)		
HC2+ non-high risk ^ь	19 (4.1)	3	3	15.8 (5.4 to 41.4)	3	15.8 (5.4 to 41.4)	3	15.8 (5.4 to 41.4)		
All HC2+	416 (90.0)	221	216	51.9 (47.2 to 56.8)	218	52.4 (47.7 to 57.3)	221	53.2 (48.4 to 58.0)		
HC2–	46 (10.0)	5	4	8.7 (3.4 to 21.5)	5	10.9 (4.7 to 24.2)	5	10.9 (4.7 to 24.2)		
All women with moderate/severe	462 (100)	226	220		223		226			

cytology

a In 12 years of follow-up.

b HC2+ but no HRHPV detected.

Women who were H	C2+ and cytolog	y negative e	entry					
			Risk from round 2					
	Number (%)	Number of CIN3+ women	2.5-year		5-year		10-year	
HPV status and cytology at round 2	of women at round 2		Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI
HC2–	806 (72.0)	2	0	0	1	0.13 (0.02 to 0.90)	2	0.26 (0.06 to 1.03)
HC2+ but no high risk detected	79 (7.1)	5	4	5.1 (1.9 to 12.9)	5	6.3 (2.7 to 14.5)	5	6.3 (2.7 to 14.5)
HRHPV+ normal cytology	158 (14.1)	14	9	5.7 (3.0 to 10.7)	11	7.0 (3.9 to 12.2)	14	8.9 (5.4 to 14.5)
Persistent HRHPV	97	13	9	9.3 (4.9 to 17.1)	11	11.3 (6.5 to 19.5)	13	13.4 (8.0 to 22.0)
New HRHPV	61	1	0	0.0	0	0.0	1	1.7 (0.2 to 11.3)
HRHPV+ borderline/ mild cytology	62 (5.5)	10	8	12.9 (6.7 to 24.2)	9	14.5 (7.8 to 26.0)	10	16.2 (9.0 to 28.0)
Persistent HRHPV	40	8	7	17.5 (8.8 to 33.2)	8	20.0 (10.5 to 36.0)	8	20.0 (10.5 to 36.0)
New HRHPV	22	2	1	4.6 (0.7 to 28.1)	1	4.6 (0.7 to 28.1)	2	9.1 (2.4 to 31.7)
HRHPV+ normal/ borderline/mild cytology	220	24						
Persistent HRHPV	137	21	16	11.7 (7.3 to 18.4)	19	13.9 (9.1 to 20.9)	21	15.3 (10.3 to 22.6)
New HRHPV	83	3	1	1.2 (0.2 to 8.3)	1	1.2 (0.2 to 8.3)	3	3.7 (1.2 to 11.0)
HRHPV+ moderate/ severe cytology	15 (1.3)	9	9	60.0 (37.2 to 83.5)	9	60.0 (37.2 to 83.5)		
Totalª	1120 (100)	40	30		34		40	

TABLE 14 Cumulative CIN3+ risks following round 2 in women who were HC2+ with negative cytology at entry (n = 1120) by HPV status at round 1 and round 2

RESULTS

a Ten further CIN3+ women were diagnosed before round 2 in cytology-negative HPV+ women in the revealed arm (see Section 4: triage of human papillomavirus-positive women, particularly the interval to retesting for human papillomavirus-positive women).

			Risk from round 2					
	Number of women at round 2	Number of CIN3+ women	2.5-year		5-year		10-year	
HPV status at round 2 (negative cytology)			Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI
Women with type-specific per	sistence							
Persistent HPV 16	41	9	7	17.1 (8.5 to 32.5)	8	19.5 (10.3 to 35.3)	9	22.0 (12.1 to 38.
Persistent HPV 18	13	1	1	7.7 (1.1 to 43.4)	1	7.7 (1.1 to 43.4)	1	7.7 (1.1 to 43.4)
Persistent 31/33/45/52/58	54	7	4	7.4 (2.9 to 18.5)	5	9.3 (4.0 to 20.8)	7	13.0 (6.4 to 25.3)
Persistent other HRHPV	26	0	0	0.0	0	0.0	0	0.0
Any HRHPV type specific persistence	134	17	12	9.0 (5.2 to 15.2)	14	10.5 (6.3 to 17.0)	17	12.7 (8.1 to 19.7)
Women without type-specific	persistence							
New HPV 16	61	2	1	1.6 (0.2 to 11.1)	2	3.3 (0.8 to 12.5)	2	3.3 (0.8 to 12.5)
New HPV 18	32	0	0	0.0	0	0.0	0	0.0
New 31/33/45/52/58	143	3	2	1.4 (0.4 to 5.5)	2	1.4 (0.4 to 5.5)	3	2.1 (0.7 to 6.4)
New other HRHPV	114	0	0	0.0	0	0.0	0	0.0
Any new HRHPV	350ª	5	3	0.9 (0.3 to 2.6)	4	1.1 (0.4 to 3.0)	5	1.4 (0.6 to 3.4)
HC2+ but no HRHPV	383	5	3	0.8 (0.3 to 2.4)	4	1.1 (0.4 to 2.8)	5	1.3 (0.6 to 3.2)
HC2-	12,028	12	5	0.04 (0.02 to 0.10)	7	0.06 (0.03 to 0.12)	12	0.10 (0.06 to 0.1
HC2 results ignoring genotypi	ng							
HC2 double positive	331	21	15	4.5 (2.8 to 7.4)	17	5.1 (3.2 to 8.1)	21	6.4 (4.2 to 9.6)
HC2 negative positive	536	6	3	0.6 (0.2 to 1.7)	5	0.9 (0.4 to 2.2)	6	1.1 (0.5 to 2.5)
HC2 positive negative	1166	2	0	0.0	1	0.09 (0.01 to 0.62)	2	0.18 (0.04 to 0.7
HC2 double negative	10,862	10	5	0.05 (0.02 to 0.11)	6	0.06 (0.02 to 0.12)	10	0.09 (0.05 to 0.1
All women	12,895	39	23		29		39	

TABLE 15 Cumulative CIN3+ risks following round 2 in women with negative cytology at round 2 by HPV status at round 1 and round 2

			Risk from round 2					
	Number of women at round 2	Number of CIN3+ women	2.5-year		5-year		10-year	
HPV status at round 2 (negative cytology)			Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI	Number of CIN3+ women (%)	95% CI
Women with type-specific pers	sistence							
Persistent HPV 16	13	7	7	53.9 (30.4 to 80.8)	7	53.9 (30.4 to 80.8)	7	53.9 (30.4 to 80.8
Persistent HPV 18	8	2	2	25.0 (6.9 to 68.5)	2	25.0 (6.9 to 68.5)	2	25.0 (6.9 to 68.5)
Persistent 31/33/45/52/58	22	3	3	13.6 (4.6 to 36.6)	3	13.6 (4.6 to 36.6)	3	13.6 (4.6 to 36.6)
Persistent other HRHPV	16	2	1	6.3 (0.9 to 36.8)	2	12.5 (3.3 to 41.4)	2	12.5 (3.3 to 41.4)
Any HRHPV type specific persistence	59	14	13	22.0 (13.4 to 34.9)	14	23.7 (14.8 to 36.8)	14	23.7 (14.8 to 36.8
Women without type-specific	persistence							
New HPV 16	42	4	4	9.5 (3.7 to 23.4)	4	9.5 (3.7 to 23.4)	4	9.5 (3.7 to 23.4)
New HPV 18	17	0	0	0.0	0	0.0	0	0.0
New 31/33/45/52/58	43	3	3	7.0 (2.3 to 20.1)	3	7.0 (2.3 to 20.1)	3	7.0 (2.3 to 20.1)
New other HRHPV	35	1	0	0.0	0	0.0	1	2.9 (0.4 to 18.6)
Any new HRHPV	137ª	8	7	5.1 (2.5 to 10.4)	7	5.1 (2.5 to 10.4)	8	5.9 (3.0 to 11.4)
HC2+ but no HRHPV	79	2	1	1.3 (0.2 to 8.7)	2	2.6 (0.6 to 9.8)	2	2.6 (0.6 to 9.8)
HC2-	304	1	1	0.3 (0.05 to 2.3)	1	0.3 (0.05 to 2.3)	1	0.3 (0.05 to 2.3)
HC2 results ignoring genotypir	ng							
HC2 double positive	140	18	15	10.7 (6.6 to 17.1)	17	12.1 (7.7 to 18.8)	18	12.9 (8.3 to 19.7)
HC2 negative positive	135	6	6	4.5 (2.3 to 10.7)	6	4.5 (2.3 to 10.7)	6	4.5 (2.3 to 10.7)
HC2 positive negative	63	1	1	1.6 (0.2 to 10.7)	1	1.6 (0.2 to 10.7)	1	1.6 (0.2 to 10.7)
HC2 double negative	241	0	0	0.0	0	0.0	0	0.0
All women	579	25	22		24		25	

TABLE 16 Cumulative CIN3+ risks following round 2 in women with borderline or mild cytology at round 2 by HPV status at round 1 and round 2

RESULTS

TABLE 17 Cumulative CIN3+ risks following round 2 by age in women with type-specific HPV persistence

		n CIN3+	Risk from round 2							
	<i>n</i> at round 2		2.5-year		5-year		10-year			
HPV status at round 2 (negative cytology)			n CIN3+ (%)	95% CI	n CIN3+ (%)	95% CI	n CIN3+ (%)	95% Cl		
23–24										
Persistent HPV 16	14	5	5	35.7 (16.7 to 65.7)	5	35.7 (16.7 to 65.7)	5	35.7 (16.7 to 65.7)		
Persistent other high-risk type	18	3	2	11.1 (2.9 to 37.6)	2	11.1 (2.9 to 37.6)	3	16.7 (5.7 to 43.2)		
25–29										
Persistent HPV 16	25	8	6	24.0 (11.6 to 45.8)	7	28.0 (14.5 to 49.9)	8	32.0 (17.5 to 53.9)		
Persistent other high-risk type	31	4	4	12.9 (5.1 to 30.8)	4	12.9 (5.1 to 30.8)	4	12.9 (5.1 to 30.8)		
30–34										
Persistent HPV 16	16	8	8	50.0 (29.0 to 75.5)	8	50.0 (29.0 to 75.5)	8	50.0 (29.0 to 75.5)		
Persistent other high-risk type	37	6	4	10.8 (4.2 to 26.3)	5	13.5 (5.9 to 29.5)	6	16.2 (7.6 to 32.6)		
35–39										
Persistent HPV 16	6	1	1	16.7 (2.5 to 72.7)	1	16.7 (2.5 to 72.7)	1	16.7 (2.5 to 72.7)		
Persistent other high-risk type	25	6	5	20.0 (8.9 to 41.6)	6	24.0 (11.6 to 45.8)	6	24.0 (11.6 to 45.8)		
≥40										
Persistent HPV 16	5	0	0		0		0			
Persistent other high-risk type	39	3	3	7.7 (2.6 to 22.0)	3	7.7 (2.6 to 22.0)	3	7.7 (2.6 to 22.0)		
All women										
Persistent HPV 16	66	22	20	30.3 (20.7 to 42.9)	21	31.8 (22.1 to 44.5)	22	33.4 (23.4 to 46.1)		
Persistent other high-risk type	150	22	18	12.0 (7.7 to 18.4)	20	13.3 (8.8 to 19.9)	22	14.7 (9.9 to 21.5)		



FIGURE 7 Cumulative CIN3+ risks following round 2 by age in women with type-specific persistence of any HRHPV.

Chapter 6 Discussion

Risk-based screening and cancer prevention

The UK National Screening Programme will adopt HPV screening and triage protocols that strike an acceptable balance between cancer risk and the cost and patient inconvenience of excessive screening and referral rates and unnecessary treatment. Acknowledging that there is no such thing as zero risk, Castle *et al.*¹⁸ proposed a risk-based strategy for cervical screening based on widely accepted practice, and suggested that women should be returned to routine screening if they have a < 2% CIN3+ risk, recalled earlier than the routine screening interval if they have a 2–10% risk, and referred to immediate colposcopy if they have a > 10% risk. These specific cut-off points may be considered arbitrary and do not consider the relationship between CIN3 and cancer risk. The aim of screening is to prevent invasive cancer and not CIN3; therefore, the acceptable cumulative CIN3+ risk when a woman is next screened depends on the ratio of cancer risk to CIN3 risk at the next screen. This ratio depends mainly on the delay between time of development of CIN3 during the screening interval and its treatment when it is detected at the next HPV test or after subsequent triage processes.

Many CIN3 cases develop soon after HPV acquisition and most develop within a few years.¹⁹ Therefore, the CIN3 incidence rate in women who were HPV– at their last test is approximately proportional to their age-specific incidence rate of new HPV infection (with a lag of a year or two), and, hence, proportional to the overall population HPV prevalence (*Figure 8*), as most prevalent infections are newly acquired and transient even in older women (see *Table 3*).

The cumulative CIN3 risk at next screen is also roughly proportional to the screening interval (*Figure 9*). The cervical cancer incidence rate in women with untreated CIN3 appears to be fairly constant at about 1% per year, independent of age,¹ so the cancer risk at next screen should be roughly proportional to the square of the screening interval. In unscreened populations with fairly constant HPV infection rates up to middle age, cervical cancer incidence should therefore rise linearly from age at first intercourse. From 10 to 30 years after first intercourse, the observed quadratic relationship corresponds closely to a linear increase with a diagnostic lag of 7.5 years.²⁰ In primary screening, women who test HPV– acquire new infections and develop resulting CIN3 throughout the interval to next test. In contrast, in the further interval before retesting when HPV+



FIGURE 8 Age-specific HRHPV prevalence at round 2 by age at round 2 overall and in women who were HRHPV– at round 1, and cumulative 10-year CIN3+ risk by age at round 1 in women who were HC2– at round 1. (HRHPV prevalence rates at round 2 by age at round 1 in women who were HRHPV– at round 1 in 5-year age groups from 20–24 years to 55–59 years were 14.7%, 8.1%, 3.6%, 2.9%, 2.2%, 1.5%, 1.5% and 0.8%.)

© Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

women are being triaged, most CIN3 lesions will have developed long before or soon after their positive HPV test. Therefore, some cancers will be present at the screening visit when HPV is detected, and the cancer risk will further increase at a roughly constant rate throughout the triage process. This simplified and speculative model ignores several relevant factors, including the lag from HPV acquisition to CIN3 development and the rate of progression to detectable cancer, but it illustrates the weakness of screening and triage protocols based only on CIN3+ risk. An undefined model of cancer risk underlies the focus on CIN3 that informs cervical screening and triage protocols. An explicit model based on epidemiological knowledge should perhaps be developed and refined as an alternative basis for cost–benefit evaluation of the details of HPV screening and triage protocols, as it is unlikely that their effects on cancer incidence can be observed directly. Even the simple comparison of HPV testing against cytology required the pooled results of trials with a total of 176,000 randomised women to detect the effect on cancer incidence.⁶

The long-term protection of a negative human papillomavirus test and, hence, the safe screening interval at different ages

The 10-year CIN3+ rate among women who were HPV– at entry to ARTISTIC is only 0.1% in those aged \geq 40 years, suggesting that screening intervals for HPV– women aged 40–49 years might be extended even further than the 6 years previously suggested.⁴ Our data are consistent with the long-term risks reported from women participating in the POBASCAM trial in the Netherlands that has led to the recommendation for the Dutch HPV screening programme to adopt a 10-year screening interval in women aged \geq 40 years.²¹ Many HPV– women have had an earlier infection, so this very low long-term risk after a single negative HPV test suggests that a HPV infection that has become undetectable rarely recurs and then progresses to CIN3. Frequent HPV testing, therefore, greatly increases the number of transient infections that are detected without increasing sensitivity for CIN3 detection.

As noted above, if most CIN3 cases (and a large proportion of CIN2 cases) persist, then their prevalence will increase almost linearly with increasing screening interval. This was seen in routinely screened women in Manchester in the 1990s, in whom CIN3 prevalence increased almost linearly with longer intervals since last normal cytology.²² Age-standardised CIN2+ prevalence in ARTISTIC shows this pattern at round 2 (*Figure 9*), with an annual increase of about 0.2% per year since entry in women with a cytologically normal entry test. *Figure 9* shows that CIN2+ prevalence at entry increased at a similar rate with increasing



FIGURE 9 Age-standardised prevalence (%) of CIN2+ during ARTISTIC by time since previous normal cytology to entry, and by time since cytologically normal entry to next test. Women whose previous cytology was abnormal or < 2 years before entry are excluded.

interval from the previous normal cytology to entry, with the addition of about 1.2% reflecting the CIN2+ prevalence missed by previous screening rounds that was detected at entry to ARTISTIC.

The role of human papillomavirus typing

The HPV+ women with borderline/mild cytology as well as those with negative cytology might be further triaged by HPV type to identify those who can be recalled later. A HPV test that identifies HPV types 16, 18 and 31/33/45/52/58 would enable this moderate-risk subgroup, which constitutes 58.4% of HPV+ women with borderline/mild cytology, to be referred immediately or followed up within (say) 1 year. See *Table 13*, which shows the 5-year CIN3+ risks of 10.9%, 4.1% and 2.9% with negative cytology, and 20.9%, 10.8% and 9.4% with borderline/mild cytology. Borderline/mild cytology with other HRHPV types (35, 39, 51, 59 and 68) has a 5-year CIN3+ risk of 3.5% and might be recalled 3 years later for a repeat HPV test. Immediate referral or early retesting of women with HPV 16 was recently recommended in the Netherlands²³ and proposed more than 10 years ago in the USA,²⁴ but the cost–benefit balance of more complex triage protocols based on full HPV typing following cytology triage deserves further consideration, particularly for vaccinated women. *Table 13* shows that most of the CIN3+ risk in HPV+ women with abnormal cytology is present at, or soon after, HPV detection and increases by about only 2% from 2.5 to 10 years irrespective of cytology.

Triage and referral policy in relation to primary screening interval

Figure 9 confirms that the marked reduction from round 1 to round 2 in the proportion of HC2+ women aged \geq 25 years who had moderate+ cytology (11.2% and 3.2%, respectively; see *Table* 6) reflects the elimination of long-standing high-grade disease that had been missed at previous routine cytology. The round 1 results, thus, represent what might be seen in primary HPV testing with a very long screening interval, whereas round 2 shows results after an interval of about 3 years. However, the overall CIN3 risks in HRHPV+ women were almost the same after stratifying by cytology, with 5-year CIN3 risks for normal, borderline/mild and moderate+ cytology of 3.1%, 9.5% and 52.4%, respectively, in round 1, and 2.5%, 8.4% and 44.7%, respectively, in round 2. This suggests that recall policy following cytology triage of HPV+ women need not depend on the primary screening interval.

Triage and referral policy in relation to type-specific human papillomavirus persistence

Table 12 shows that the 10-year CIN3 risk for women with a new HPV type at round 2 was only 3.4% (95% CI 2.1% to 5.4%). The risk was no higher in those whose HPV infection at round 1 had cleared (2.7%, 95% CI 0.9% to 8.0%) than in those who were HPV– at round 1 (3.6%, 95% CI 2.2% to 6.0%). In contrast, the 10-year CIN3 risk in women with type-specific persistence at round 2 is six times greater than for a new infection of the same type. Among 331 women with HRHPV at both rounds, 58% had the same type (10-year CIN3 risk 20.4%, 95% CI 15.6% to 26.4%). The 10-year CIN3 risk in the remaining 42% who had cleared the original type and acquired a new type was only 2.7%, so these women could safely be recalled 3 years later when most of their infections will have cleared. This large difference in risk (p < 0.0001), and, hence, the potential efficiency of further triage, is diluted with HPV tests that do not identify all 13 high-risk types. The price of different HPV tests and the disadvantages of more complex triage and referral protocols will determine whether or not the advantages of full HPV typing outweigh the costs.

A triage protocol in which women are recalled repeatedly until their CIN3 risk exceeds (say) 10% implies a cycle of repeated testing that ends only in HPV clearance or referral, and some limit must be set to detect and treat occult CIN3 that has regressed cytologically. A HPV infection that has persisted for 5 years should

[©] Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

perhaps be referred for random cervical biopsy. In a prospective study in the Netherlands, six out of the eight women with untreated CIN who regressed to normal cytology and colposcopy but remained HRHPV+ over several years had occult CIN3 detected by random biopsy.²⁵

We excluded women with any recorded histology in calculating HPV clearance rates, as biopsy can induce clearance.²⁶ This might lead to an overestimate of clearance but, as the clearance rates for new and persistent HRHPV infections shown in *Table 4* are similar to those reported by Nobbenhuis *et al.*²⁷ in women followed up without intervention, any bias seems to be small.

Triage of human papillomavirus-positive women

If the screening interval is increased, the time between appearance and diagnosis of CIN3 and, hence, the risk of progression to cancer will also increase, so the triage protocol must minimise the delay in referring those at highest risk to colposcopy. Nonetheless, the evidence from round 1 of ARTISTIC (see Table 13) indicates that the triage protocol being piloted in England could be too conservative, at least for HPV+ women with negative cytology. The risk of developing CIN3 within 5 years in HPV+ women is 52.4% for moderate/severe cytology, 9.6% for borderline/mild cytology and only 3.1% for normal cytology (see Table 13). The suggestion that a CIN3+ risk of about 10% is a reasonable threshold for immediate colposcopy^{18,28} would, therefore, imply rapid referral for HPV+ women with any abnormal cytology except perhaps the minority of women with borderline/mild cytology who have genotypes with the lowest associated risks. The minimal increase in the risk of developing cancer by retesting women with borderline/ mild cytology rather than referring them directly to colposcopy has always been accepted, and it is not clear that this risk/benefit balance should be abandoned when they are found to have HPV. In round 1 of ARTISTIC, the proportion of HPV+ women with moderate + cytology was 12% below age 50 years and at age 50 years or older. By round 2 of ARTISTIC, only 5% of HPV+ women aged < 50 years and 1% of women aged \geq 50 years had moderate/severe cytology. The majority of HPV+ women in a population regularly screened by HPV testing will, thus, have negative or borderline/mild cytology.

Triage: women with normal cytology

Of the women who were HC2+ with normal cytology at entry to ARTISTIC, 72% had cleared their HPV infection within 3 years (see *Table 14*). Only 1.3% (15 out of 1116) of women had moderate or worse cytology by round 2 (of whom, nine were diagnosed with CIN3). Fourteen per cent of women remained HPV+ with normal cytology, but approximately 60% of these (97 out of 158) had type persistent HPV and approximately 40% (61 out of 158) had new HPV types. The subsequent elevated cumulative risk was among those with type-specific infections. This also applied to those with borderline/mild cytology at the follow-up, but this was based on small numbers. This implies that the current policy in the pilot sites (repeat testing after 1 year and referral to colposcopy after 2 years if women are still HPV+ with normal cytology) is referring a large proportion of women to unnecessary testing and colposcopy. Recall after 2 or 3 years then referring only those developing moderate+ cytology and those with type-specific persistence may be a better strategy.

Triage: women with borderline/mild cytology

Among HPV+ 24- to 64-year-old women presenting with borderline/mild cytology at entry, 29% had cleared their HPV infection within 1 year and an additional 19% remained HPV+ but became cytologically normal. Those with HPV 16 with 5-year CIN3+ risk of 20.5% should clearly be referred immediately, but inviting all HPV+ women with borderline or mild cytology to return for a repeat test at 1 year would substantially reduce immediate referrals and could be a better strategy.

Ages at starting and stopping screening

Stopping screening

The effect on long-term CIN3+ risk of stopping screening in women who are HPV- at various ages can be inferred from Table 10. For example, the 10-year CIN3+ risk in HPV– women aged \geq 50 years is only 0.1%. Their risk of acquiring HPV is an order of magnitude lower than for women aged 20–24 years (see Table 3), and there is plausible epidemiological evidence that HPV infection in middle age confers little lifetime risk of cervical cancer.²⁰ Conversely, the case for a further test at least 10 years later is that their lifetime cancer risk could also be of the order of 1 per 1000 and possibly higher. The sharp decline with increasing age in the CIN3-to-cancer ratio in ARTISTIC (see Table 8), and the observation that CIN3 that remains detectable by HPV testing and random biopsy can regress to become cytologically and colposcopically undetectable²⁵ suggests that their CIN3 risk may be underestimated. A quirk of the current screening protocol is that a woman's age at her final test ranges from 60 to 64 years because the final invitation is 5 years after her previous test unless this was at age 60 years or older. One final test at the age of 65 years could be offered to all women who were HPV- at the age of 50 years. Whatever the age at stopping, the final HPV test might require a more sensitive test than at younger ages. Five out of the six women in ARTISTIC who developed invasive cervical cancer after a negative HC2 entry test were found to be HRHPV+ when the stored entry sample was retested by PCR. Alternatively, HPV and cytology cotesting might be considered at a woman's final test. The one woman in ARTISTIC and all three in the POBASCAM trial²¹ who developed invasive cancer within 4 years of a HPV- entry test were cytologically abnormal at entry.

Age at starting screening

The natural history of HPV infection, persistence and progression to CIN3 is similar at all ages from 20 years to about 45 years, after which the CIN3 rate in HPV+ women falls (see *Figure 7*) and cervical cancer incidence in unscreened populations shows a sharp inflection.²⁰ Therefore, the age at which screening begins (currently 25 years) might be reconsidered when HPV testing is introduced. Owing to this, we have outlined some of the effects of beginning HPV screening at (say) age 22 years to illustrate the issues involved. Screening after age 25 years reduces cervical cancer incidence, particularly for more advanced disease;² therefore, earlier screening would presumably achieve some reduction in the rising cancer incidence in young women. The effect on referrals to colposcopy would depend on whether immediate referral was restricted to women with moderate or severe cytology, and on the recall interval for HPV+ women with normal, borderline or mild cytology. Almost 40% of women aged 20–24 years were HC2+ at entry to ARTISTIC, but only 4.2% had moderate or severe cytology.³ A second test at age 25 years for those who were HPV+ at age 22 years would enable type-specific persistence to be identified and triaged. As discussed above, the CIN3+ risk is much lower in those with a new HPV infection; therefore, they could safely be recalled 3 years later, when most infections will have cleared, except perhaps those with HPV 16 (see *Table 12*).

Study limitations

The effect of cervical sampling and biopsy on human papillomavirus clearance

A potential bias in the observed 10-year cumulative CIN3+ rates in HPV– women is the possibility that intervening screening, biopsy and treatment could have induced clearance of subsequent HPV infections and prevented progression to CIN3.²⁶

Variation in cytology practice

ARTISTIC recruited from 127 centres spread throughout the Greater Manchester area and cytology was performed in two laboratories where the staff were retrained ahead of the national roll-out of LBC, which improved the sensitivity of our cytology,¹⁰ as this varies between laboratories. The HC2 test used in ARTISTIC has largely been replaced for use in the NHS by modern PCR-based assays. We do not know how these assays would have performed, particularly in detecting HPV among women who tested HC2– at entry but subsequently developed invasive cervical cancer (see *Table 7*). Despite these potential issues, we have shown similarities between the first round of ARTISTIC and the national pilot study (see *Table 6*) and believe the results to be generalisable to the British population.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: INIRH Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Hybrid capture 2 test sensitivity

Table 2 shows that about 4% of samples from women aged ≥ 25 years who tested positive for HPV by HC2 showed no HRHPV on genotyping. These constituted over half of HC2+ results in women aged ≥ 50 years. A total of 40% of these women had a RLU of between 1 and 2 pg/µl, indicating that most are probably false positives.¹¹ (The definition of HC2 test positivity is a RLU of ≥ 1 pg/µl.) Our conclusions are unlikely to have been much affected by ignoring these women who were HC2+ but negative for the 13 high-risk target types (4.2% of all women in round 1 and 3.5% in round 2; see *Table 2*). Their 10-year CIN3 rate was 1.8% (3.0% in those with low-risk HPV types and 0.8% in those negative for all 27 HPV types identified by the linear array), and their inclusion in the HC2– group would increase the 10-year CIN3+ risk in HC2– women from 0.31% to 0.37%. Low-risk HPV types occasionally cause high-grade neoplasia but by definition are not associated with invasive cancer risk, so the proportion of high-grade CIN cases that are missed by HRHPV testing is not an indication of inadequate sensitivity. Modern PCR-based HRHPV tests may have slightly better sensitivity with respect to cancer than the HC2 test.

Human papillomavirus-negative cancers

The small proportion of cervical cancers that do not contain HPV DNA will be missed by any HPV-based primary test. Only 0.3% were HPV– in an international series of cancer biopsies with stringent criteria for diagnosis and PCR adequacy and separate PCR assays for each high-risk type.²⁹ A more relevant estimate for the practical purpose of improving routine screening is the proportion of cancers in which HPV cannot be detected in recent cervical samples. None of the latest available samples before diagnosis from the 23 cancers in this cohort was HPV– on retesting, but *Table 7* shows that the entry sample was HC2 negative but HPV positive by PCR in 5 (22%). Larger numbers tested by standard modern assays will provide a more precise estimate of this important index of sensitivity.

Further research

The National Institute for Health Research has funded a further extension, under which follow-up for CIN2 diagnoses (which are managed clinically in the same way as CIN3 but are not recorded centrally) will be updated through local histology laboratories. The sample bank, augmented by further samples from women in the cohort who are screened between 2018 and 2022, will be used to retest samples from women who develop CIN3 or cancer and matched controls. Interim results for CIN2+ as well as CIN3+ will be provided to the NSC in order to inform its recommendations on screening intervals for HPV– women and triage protocols following HPV detection. The same data would be available in the future from national routine screening records if these were linked to histology records to capture CIN2 diagnoses as well as to national cancer registration. The establishment of a larger national bank of samples from routinely screened women would provide much larger numbers of samples from women who will develop cervical cancer for retesting by more sensitive PCR and with alternative screening or triage assays such as DNA methylation.

Conclusions

A screening interval of up to 10 years following a negative HPV test might be considered for women aged \geq 40 years and perhaps for younger women following a negative HPV test. Immediate referral of all HPV+ women with mildly abnormal cytology, as adopted in the English pilot study of primary HPV screening, may not be essential and recall intervals of 1 year for high-risk subgroups and 3 years for the remainder could be considered. The risk is largely confined to women with a persisting type-specific HPV and a 10-year interval will identify them efficiently. The possibility of a small increase in risk of progression to invasive cancer between screening rounds and perhaps some increase in the proportion of women who fail to attend regularly must be weighed against the advantages for patients and reduction of NHS costs with longer screening intervals.

Routine HPV testing and follow-up to HPV clearance will divide the population into the low-risk HPV– majority and those with type-specific persistent HPV infection among whom virtually all cervical cancers will arise. Those whose infections do not disappear after repeated testing may consider curative treatment irrespective of cytological or colposcopic findings, particularly when they reach the age at which routine screening ends.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Acknowledgements

We would like to thank the original ARTISTIC trial study group (listed below), Professor Attila Lorincz and Dr Caroline Reuter at the Wolfson Institute for testing stored samples from women who subsequently developed invasive cervical cancer, and the women who participated in the trial.

The ARTISTIC trial study group

Chief investigators

HC Kitchener (Clinical Principal Investigator; University of Manchester).

J Peto (Epidemiological Principal Investigator; London School of Hygiene & Tropical Medicine).

Trial co-ordinators

- P Wheeler (University of Manchester).
- C Thomson (University of Manchester).

R Albrow (University of Manchester).

Epidemiology/statistics

C Gilham (London School of Hygiene & Tropical Medicine).

M Almonte (London School of Hygiene & Tropical Medicine).

C Roberts (University of Manchester).

S Moss (Institute of Cancer Research).

Cytopathology

M Desai (Manchester Cytology Centre).

J Mather (Manchester Cytology Centre).

Virology

A Sargent (Department of Virology, Central Manchester University Hospitals).

A Bailey (Department of Virology, Central Manchester University Hospitals).

A Turner (Department of Virology, Central Manchester University Hospitals).

Contributions of authors

Ms Clare Gilham (Medical Statistician) carried out the data management and statistical analysis and drafted the report.

Dr Alexandra Sargent (Clinical Scientist) conducted the virology testing for ARTISTIC.

[©] Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Professor Henry C Kitchener (Professor of Gynaecological Oncology) was the clinical principal investigator for ARTISTIC and made critical comments on the report.

Professor Julian Peto (Professor of Epidemiology) contributed to the design of the study and the analysis and drafted the report.

Data-sharing statement

All data requests should be submitted to the corresponding author for consideration. Access to available anonymised data may be granted following review.

Patient data

This work uses data provided by patients and collected by the NHS as part of their care and support. Using patient data is vital to improve health and care for everyone. There is huge potential to make better use of information from people's patient records, to understand more about disease, develop new treatments, monitor safety, and plan NHS services. Patient data should be kept safe and secure, to protect everyone's privacy, and it's important that there are safeguards to make sure that it is stored and used responsibly. Everyone should be able to find out about how patient data are used. #datasaveslives You can find out more about the background to this citation here: https://understandingpatientdata.org.uk/data-citation.

References

- 1. Peto J, Gilham C, Fletcher O, Matthews FE. The cervical cancer epidemic that screening has prevented in the UK. *Lancet* 2004;**364**:249–56. https://doi.org/10.1016/S0140-6736(04)16674-9
- Landy R, Pesola F, Castañón A, Sasieni P. Impact of cervical screening on cervical cancer mortality: estimation using stage-specific results from a nested case-control study. *Br J Cancer* 2016;**115**:1140–6. https://doi.org/10.1038/bjc.2016.290
- Kitchener HC, Almonte M, Gilham C, Dowie R, Stoykova B, Sargent A, et al. ARTISTIC: a randomised trial of human papillomavirus (HPV) testing in primary cervical screening. *Health Technol Assess* 2009;**13**(51). https://doi.org/10.3310/hta13510
- Kitchener HC, Gilham C, Sargent A, Bailey A, Albrow R, Roberts C, et al. A comparison of HPV DNA testing and liquid based cytology over three rounds of primary cervical screening: extended follow up in the ARTISTIC trial. Eur J Cancer 2011;47:864–71. https://doi.org/10.1016/j.ejca.2011.01.008
- Kitchener HC, Canfell K, Gilham C, Sargent A, Roberts C, Desai M, Peto J. The clinical effectiveness and cost-effectiveness of primary human papillomavirus cervical screening in England: extended follow-up of the ARTISTIC randomised trial cohort through three screening rounds. *Health Technol Assess* 2014;**18**(23). https://doi.org/10.3310/hta18230
- Ronco G, Dillner J, Elfström KM, Tunesi S, Snijders PJ, Arbyn M, et al. Efficacy of HPV-based screening for prevention of invasive cervical cancer: follow-up of four European randomised controlled trials. *Lancet* 2014;**383**:524–32. https://doi.org/10.1016/S0140-6736(13)62218-7
- Kitchener HC. HPV primary cervical screening: time for a change. Cytopathology 2015;26:4–6. https://doi.org/10.1111/cyt.12236
- Kitchener HC, Almonte M, Wheeler P, Desai M, Gilham C, Bailey A, et al. HPV testing in routine cervical screening: cross sectional data from the ARTISTIC trial. Br J Cancer 2006;95:56–61. https://doi.org/10.1038/sj.bjc.6603210
- Sargent A, Bailey A, Almonte M, Turner A, Thomson C, Peto J, et al. Prevalence of type-specific HPV infection by age and grade of cervical cytology: data from the ARTISTIC trial. Br J Cancer 2008;98:1704–9. https://doi.org/10.1038/sj.bjc.6604324
- Kitchener HC, Almonte M, Thomson C, Wheeler P, Sargent A, Stoykova B, et al. HPV testing in combination with liquid-based cytology in primary cervical screening (ARTISTIC): a randomised controlled trial. Lancet Oncol 2009;10:672–82. https://doi.org/10.1016/S1470-2045(09)70156-1
- Sargent A, Bailey A, Turner A, Almonte M, Gilham C, Baysson H, et al. Optimal threshold for a positive hybrid capture 2 test for detection of human papillomavirus: data from the ARTISTIC trial. J Clin Microbiol 2010;48:554–8. https://doi.org/10.1128/JCM.00896-09
- Almonte M, Silva Idos S, Asare A, Gilham C, Sargent A, Bailey A, et al. Sexual behavior and HPV infection in British women, by postal questionnaires and telephone interviews. J Med Virol 2011;83:1238–46. https://doi.org/10.1002/jmv.22085
- Crosbie EJ, Bailey A, Sargent A, Gilham C, Peto J, Kitchener HC. The PapilloCheck assay for detection of high-grade cervical intraepithelial neoplasia. J Clin Microbiol 2015;53:3553–9. https://doi.org/10.1128/JCM.01578-15

[©] Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

- Veldhuijzen NJ, Berkhof J, Gillio-Tos A, De Marco L, Carozzi F, Del Mistro A, et al. The age distribution of type-specific high-risk human papillomavirus incidence in two population-based screening trials. Cancer Epidemiol Biomarkers Prev 2015;24:111–18. https://doi.org/10.1158/ 1055-9965.EPI-14-0628
- 15. Moss S, Gibney A. *HPV Primary Screening Pilots: Evaluation Report to the National Screening Committee*. 2015. URL: https://legacyscreening.phe.org.uk/cervicalcancer (accessed 19 April 2018)
- Ronco G, Giorgi-Rossi P, Carozzi F, Confortini M, Dalla Palma P, Del Mistro A, et al. Efficacy of human papillomavirus testing for the detection of invasive cervical cancers and cervical intraepithelial neoplasia: a randomised controlled trial. *Lancet Oncol* 2010;**11**:249–57. https://doi.org/10.1016/ S1470-2045(09)70360-2
- Luyten A, Buttmann-Schweiger N, Luyten K, Mauritz C, Reinecke-Lüthge A, Pietralla M, *et al.* Early detection of CIN3 and cervical cancer during long-term follow-up using HPV/Pap smear co-testing and risk-adapted follow-up in a locally organised screening programme. *Int J Cancer* 2014;**135**:1408–16. https://doi.org/10.1002/ijc.28783
- Castle PE, Sideri M, Jeronimo J, Solomon D, Schiffman M. Risk assessment to guide the prevention of cervical cancer. Am J Obstet Gynecol 2007;197:356.e1–6. https://doi.org/10.1016/j.ajog.2007.07.049
- Schiffman M, Castle PE, Jeronimo J, Rodriguez AC, Wacholder S. Human papillomavirus and cervical cancer. *Lancet* 2007;**370**:890–907. https://doi.org/10.1016/S0140-6736(07)61416-0
- Plummer M, Peto J, Franceschi S, International Collaboration of Epidemiological Studies of Cervical Cancer. Time since first sexual intercourse and the risk of cervical cancer. *Int J Cancer* 2012;**130**:2638–44. https://doi.org/10.1002/ijc.26250
- 21. Dijkstra MG, van Zummeren M, Rozendaal L, van Kemenade FJ, Helmerhorst TJ, Snijders PJ, *et al.* Safety of extending screening intervals beyond five years in cervical screening programmes with testing for high risk human papillomavirus: 14 year follow-up of population based randomised cohort in the Netherlands. *BMJ* 2016;**355**:i4924. https://doi.org/10.1136/bmj.i4924
- 22. Peto J, Gilham C, Deacon J, Taylor C, Evans C, Binns W, *et al.* Cervical HPV infection and neoplasia in a large population-based prospective study: the Manchester cohort. *Br J Cancer* 2004;**91**:942–53. https://doi.org/10.1038/sj.bjc.6602049
- 23. Ebisch RM, de Kuyper-de Ridder GM, Bosgraaf RP, Massuger LF, IntHout J, Verhoef VM, *et al.* The clinical value of HPV genotyping in triage of women with high-risk-HPV-positive self-samples. *Int J Cancer* 2016;**139**:691–9. https://doi.org/10.1002/ijc.30090
- Khan MJ, Castle PE, Lorincz AT, Wacholder S, Sherman M, Scott DR, et al. The elevated 10-year risk of cervical precancer and cancer in women with human papillomavirus (HPV) type 16 or 18 and the possible utility of type-specific HPV testing in clinical practice. J Natl Cancer Inst 2005;97:1072–9. https://doi.org/10.1093/jnci/dji187
- Nobbenhuis MA, Helmerhorst TJ, van den Brule AJ, Rozendaal L, Voorhorst FJ, Bezemer PD, et al. Cytological regression and clearance of high-risk human papillomavirus in women with an abnormal cervical smear. Lancet 2001;358:1782–3. https://doi.org/10.1016/S0140-6736(01)06809-X
- Koss LG, Stewart F, Foote FW, Jordan MJ, Bader GM, Day E. Some histological aspects of behavior of epidermoid carcinoma in situ and related lesions of the uterine cervix. a long-term prospective study. *Cancer* 1963;**16**:1160–211. https://doi.org/10.1002/1097-0142(196309)16:9<1160::AID-CNCR2820160910>3.0.CO;2-4
- Nobbenhuis MA, Walboomers JM, Helmerhorst TJ, Rozendaal L, Remmink AJ, Risse EK, et al. Relation of human papillomavirus status to cervical lesions and consequences for cervical-cancer screening: a prospective study. *Lancet* 1999;**354**:20–5. https://doi.org/10.1016/S0140-6736(98) 12490-X

- 28. Wentzensen N, Schiffman M, Palmer T, Arbyn M. Triage of HPV positive women in cervical cancer screening. *J Clin Virol* 2016;**76**(Suppl. 1):49–55. https://doi.org/10.1016/j.jcv.2015.11.015
- 29. Walboomers JM, Jacobs MV, Manos MM, Bosch FX, Kummer JA, Shah KV, *et al.* Human papillomavirus is a necessary cause of invasive cervical cancer worldwide. *J Pathol* 1999;**189**:12–19. https://doi.org/10.1002/(SICI)1096-9896(199909)189:1<12::AID-PATH431>3.0.CO;2-F

© Queen's Printer and Controller of HMSO 2019. This work was produced by Gilham *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

EME HS&DR HTA PGfAR PHR

Part of the NIHR Journals Library www.journalslibrary.nihr.ac.uk

This report presents independent research funded by the National Institute for Health Research (NIHR). The views expressed are those of the author(s) and not necessarily those of the NHS, the NIHR or the Department of Health and Social Care

Published by the NIHR Journals Library